



# Article Home-Based Intelligent Exercise System for Seniors' Healthcare: The Example of Golf Croquet

Yu-Li Zhou<sup>1</sup>, Chia-Li Chen<sup>2</sup>, Shih-Jui Chang<sup>3,\*</sup> and Bo-Sheng Wu<sup>2</sup>

- <sup>1</sup> Taiwan Economy and Industry Association, Taipei 100008, Taiwan; olivia.chen@hyiasset.com
  - Department of Information Management, Lunghwa University of Science and Technology,
  - Taoyuan 333326, Taiwan; carey71@gm.lhu.edu.tw (C.-L.C.); qwas90101@gmail.com (B.-S.W.)
- <sup>3</sup> Center for Professional Competency Certification, Lunghwa University of Science and Technology, Taoyuan 333326, Taiwan
- \* Correspondence: shihjui.chang@gmail.com; Tel.: +886-2-82093211

Abstract: In 2020, the world experienced the threat of the COVID-19 epidemic, and seniors and chronic disease patients generally reduced their exercise and social activities to avoid increasing the risk of infection, which could lead to increased loneliness and even many diseases. Combining golf croquet games with a mobile application (App) and AIoT companion robots, this research constructs a home-based intelligent exercise system, uses the technology acceptance model (TAM), deduces users' intention to use this system based on perceived usefulness and perceived ease of use, and adds the needs of love and belonging, esteem, cognitive, aesthetic, and self-actualization in Maslow's hierarchy of needs theory (HNT) to conduct an analysis of system needs. This study collected empirical data, totaling 320 participants including seniors and their caregivers, from elderly care centers in northern Taiwan using a cross-sectional survey and purposive sampling. Based on regression and variance analysis, the results show that participants have a high level of acceptance of this system, believing that it is easy to learn and operate and can increase interaction with others, improve self-confirmation, satisfy the thirst for knowledge, increase the feeling of happiness, and fulfill self-actualization needs. In the future, by collecting and recording the process of seniors using the App, so as to find out their health problems as soon as possible, expand their daily life through this exercise, and achieve the goal of happy living and better healthcare.

**Keywords:** technology acceptance model; Maslow's hierarchy of needs; golf croquet; mobile applications; companion robot; AIoT; senior

# 1. Introduction

With economic advancement and the progress of science, technology, and medical treatments, human life expectancy is continuing to grow. Indeed, the global aging population is increasing year by year, which has become a topic of close attention by many countries [1–3]. With the change in society and the change in the concept of procreation in Taiwan, its population growth has slowed down, and its national fertility rate has decreased year by year. In addition, the prevalence of late marriage and the rise in average life expectancy have significantly changed the age structure of Taiwan's population [4]. The social problems of aging span not only the needs of the elderly's welfare, but also economic, medical, and family issues. Secondary issues relate to the elderly's life, leisure and entertainment, well-being, and psychological and social adaptation, which the government and society should pay attention to and properly plan for [5–7].

For strong and rich countries, the greatest risk for seniors is not entirely health problems but also isolation and loneliness [8–10]. For seniors, regular exercise is the most effective way to reduce organ degradation and improve physical aging. By participating in leisure sports or activities that they are interested in and obtaining happiness from



Citation: Zhou, Y.-L.; Chen, C.-L.; Chang, S.-J.; Wu, B.-S. Home-Based Intelligent Exercise System for Seniors' Healthcare: The Example of Golf Croquet. *Sports* **2023**, *11*, 207. https://doi.org/10.3390/ sports11110207

Academic Editors: Wenfei Zhu and Yuliang Sun

Received: 13 August 2023 Revised: 14 September 2023 Accepted: 24 October 2023 Published: 26 October 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). them, people can achieve physical and mental satisfaction and avoid isolation and loneliness [11,12]. If seniors actively participate in leisure activities that are beneficial to their body and mind and recognize the significance and importance of leisure to quality of life, then they can successfully move toward aging and reduce their social burden [13,14].

Among many sports available to most people, golf croquet is a low-intensity sport and a leisure activity with both fitness and social functions. Seniors typically have plenty of time to participate in croquet, which has a great effect on physical and mental health [15]. In addition to gaining recognition among peers, they can temporarily get rid of unhappiness and become happy from this activity, and depression and anxiety can be effectively relieved. To enable seniors to use it at home, it is necessary to improve the limitations of the open and flat field and the assistance of many staff members needed in the game of golf croquet. The present study develops a golf croquet scoring system to optimize this game and displays and calculates the score and ranking using a mobile application (App). The system combines Internet of Things technology, a pressure sensor, and a Zenbo robot, and its patent was applied for in the Republic of China (Patent No. TWM602930). To explore a sports system suitable for seniors to use at home, this study takes golf croquet as an example and proposes a revised hierarchy of needs theory (HNT) and technology acceptance model (TAM) to confirm seniors' demand for a healthy home exercise system.

In addition to anticipating the convenience of life and the development of civilization brought by new technology, people may also experience exclusion due to the uncertainty and unknown risks of technological trends. Many scholars have cited the TAM to enhance users' willingness to use new technologies [16–19]. As proposed by Davis et al. [20], TAM was designed for users' behavior in accepting new information systems while analyzing various factors that affect user acceptance. It also provides a theoretical basis for understanding the impact of external factors on the beliefs, attitudes, and intentions of users' internal factors and then on their use of technology.

People's desires and impulses are interrelated, and what drives human beings is a number of constant, hereditary, and instinctive needs. These needs are not only physiological but also psychological. They are inherent in human nature, and culture cannot stop them but only restrain them. Scholars have cited HNT to discuss people's internal needs for things that then influence their behavioral motives [21–24]. Proposed by Maslow [25], the HNT is a well-known motivational theory and a classification system aimed at reflecting the general needs of society as its foundation and subsequently developing into more acquired emotions. Starting from the lowest level of physiological needs, when one level of needs is met, one will then pursue the higher level of needs of the next level until reaching super self-actualization [26].

This study retains perceived usefulness and ease of use in the technology acceptance model and adds Maslow's demand theory to find out what factors elderly people think will affect their willingness to use a golf croquet exercise system, which can be taken as a reference framework for developing a seniors' sports management system in the future. An investigation of the golf croquet home exercise system in this study further explores the testers' acceptance of the functional orientation and demand design of the system. In the future, relevant information during the use process can be collected and recorded, can be used to detect the health problems of users early, and can promote this research to penetrate the daily lives of seniors.

The rest of the paper is structured as follows. Section 2 provides a literature review. Section 3 introduces the methodology used herein, including Maslow's needs theory to improve the TAM, the questionnaire design, and the analysis methods. Section 4 describes the data collection and analysis of the results, including sample description, reliability and validity analysis, hypothesis verification, and discussion. Section 5 presents the conclusion. Section 6 offers the contribution and applicability of the study, and the limitations are stated in Section 7.

#### 2. Literature Review and Research Hypotheses

#### 2.1. The Needs and Characteristics of Seniors

Population aging is a global problem, and Taiwan is no exception. Due to the substantial improvement in living standards and the continuous progress of medical technology, science, and technology, people's awareness of health and wellness is also gradually rising. According to data from a survey report on the elderly by the Ministry of Health and Welfare of Taiwan [27], about 71% of the seniors feel that their health and physical and mental functions are at the middle level—that is, most seniors are still in a relatively healthy state. Therefore, when seniors have relatively more time at their disposal, the arrangement of daily life not only can replace their work style before retirement but also become an important key to maintaining quality of life. In other words, the arrangement of various leisure and health activities is more important for seniors, and this changing trend is also of wide concern to many in academic circles [28,29].

According to data from that same survey report on seniors [27], about 76% of Taiwan's 55-year-old citizens believe that their physical and mental health is in an ordinary or even good condition. The statistical data also show that "health care group activities" have the highest participation rate, and "leisure and entertainment group activities" are also those they often participate in, regardless of whether the elderly population is aged 55 to 64 or aged 65 or older. This means that Taiwan's seniors have a strong demand for leisure and healthcare activities. On the other hand, the proportion of seniors over 65 who need nursing, maintenance, or care services is only 17%. Although the physiological function of elderly people will inevitably decline, their psychological pursuit of respect, attention, and companionship will not disappear with age but will continue to increase [30]. With the progress of medical treatment and technology, the maintenance of elderly people's physiological functions is also improving. It is expected that the number of elderly people over 80 years old in Taiwan's elderly population will continue to increase.

#### 2.2. Recreational Sport and Golf Croquet

Leisure comes from the Latin word "Licere", which means free time or free activities after getting rid of productive labor [31]. Leisure sports refer to those in which participants freely choose to engage for their own fun during leisure time. Such sports include physical exercise and recreational sports [32,33]. Leisure sports emphasize personal will and self-development, the pursuit of health and physical development, and achieving the purpose of relaxation. More importantly, one can obtain internal satisfaction from sports [34].

With the transformation of society, machines are replacing human beings in various tasks, the number of working hours of human beings has gradually decreased, and the amount of free time has increased. Therefore, how to arrange appropriate rest and life outside work has increasingly attracted public attention. After entering the elderly age group, individuals mainly participate in physical leisure activities (such as aerobics or walking), intellectual leisure activities (such as chess), religious leisure activities (such as charitable activities), and social activities (such as volunteering). If the number of leisure, recreational, educational, and sports courses can be increased and some opportunities can be provided to participate in voluntary services for seniors, then this will fill their spare time and give them the opportunity to participate in society. The most important thing is to help them establish a correct self-awareness so that they will not think that they have lost their usefulness in society [35].

Activity or exercise forms a health guide designed for the human body. If an individual does not engage in a certain degree of activity or exercise, then muscle strength will gradually decline with both age and the aging phenomenon, resulting in a drop in activity, which will increase the possibility of accidents in seniors and relatively increase the risk of chronic diseases [36]. Croquet is very suitable for seniors to participate in and reduce various chronic diseases [15,37]. For them, participation in croquet can help develop interpersonal relationships and interactions after retirement. It is a low-intensity sport, has fitness and social functions, and is quite suitable for the elderly. By participating in croquet,

seniors can gain the recognition of their peers and temporarily get rid of unhappiness with this game. Depression and anxiety can thus be effectively relieved [35,38].

Croquet originated in France. It is a recreational activity that can be participated in by many people, and it can help with mind power [39]. However, it is also subject to some restrictions, including limitations in field size. Golf croquet originated in Japan and drew lessons from the traditional British croquet sport that rose in the 14th century. At that time, it provided children with a means of entertainment, but it was not accepted by them; instead, it was loved by seniors. It is an indoor sport aptly suitable for seniors [40,41].

Golf croquet combines fun and brainstorming. It integrates the training of feet, hands, and concentration into an activity so that participants can also stand around, chat, and walk happily and naturally. Golf croquet requires more attention and better control, and the rich colors on the scoring carpet improve the willingness to participate. The score matrix in Figure 1 is 13 grids, indicating different scores. The score is doubled when the ball falls at the selected number. Scores can also be negative. The arrangement of competition activities can make everyone more aware of the importance of cooperation and fully open the door to group activities where people are able to laugh and enjoy time with others.



Figure 1. Schematic diagram of golf croquet.

#### 2.3. Internet of Things and Companion Robot

Taiwan's society is facing the phenomenon of aging and fewer children, causing a hidden worry about the country's sustainable development. How to guarantee the quality of life of seniors and reduce the burden of young people are important issues for society.

From the perspectives of science and technology, the maturity of IoT technology, and the rapid development of AI, the use of AIoT emerging technology combines AI and IoT to improve the quality of life of seniors and has attracted greater attention [42–46]. IoT means that electronic devices can be connected to the Internet through network infrastructure [47]. The device has a variety of sensors, which can collect and integrate information anytime and anywhere to assist people's lives [48,49]. In recent years, IoT has been used more and more widely, covering a variety of different application fields. Among them, Qian et al. [50] used ambient assisted living (AAL) and a healthcare monitor (HM) in combination with AI and IoT to systematically help seniors live a more relaxed and better life in terms of methodology and application scenarios. Abdi et al. [51] found that some emerging technologies (such as AIoT) may solve the elderly's basic self-care and medical care needs, but there is still a gap in their potential use in some nursing fields. Research results have provided the industry with some clear challenges and obstacles to improving the adoption of emerging technologies by the elderly, such as robots and virtual/augmented reality (AR/VR). The integration of IoT and mobile phone devices can effectively solve the difficulties of limited storage space and insufficient computing processing capacity of devices at low power consumption, strengthen the development of the entire IoT technology

space, and achieve effective management in terms of reliability, computing performance, security, and confidentiality [43,45,52,53].

With the progress of science and technology, people expect robots to be close to human life and provide more diversified services. Various service robots have begun to flourish, which try to integrate into human life with different roles. In elderly care, the development of many machine pets and machine dolls, which can interact with the elderly, really helps ease their psychological loneliness and lets them feel the value of being needed again [54]. Companion robots have evolved from mental commitment robots to therapeutic robots that provide psychological and spiritual healing effects for the elderly, relieve their stress and loneliness, offer psychological healing functions for patients with dementia, and improve their mental health and quality of life [55]. Studies [56–60] also showed that if you want to improve the quality of life of seniors, then you can use home-based robots to achieve the goal and also effectively reduce the burden of caregivers.

The rise of AI in recent years, especially natural language processing technology, has gradually matured and become widespread. Voice assistants combined with IoT technology have already entered many homes. Users can use their voice to perform functional tasks or information queries through a voice assistant, which can even be applied to the biomedical field, including disease diagnosis and life assistance [61]. Many research and development projects also try to combine robots and natural language processing to develop robots that can talk naturally. For example, Zenbo (as shown in Figure 2), a robot developed by ASUS, is a companion robot that combines the cloud natural language processing engine. Our work uses ASUS Zenbo to connect to our App. Zenbo Software Development Kit and additional tools can accelerate our development of other Apps and capabilities for Zenbo.



Figure 2. Zenbo robot.

Because Zenbo has a variety of functions, it can cover the basic needs of seniors in daily life and company [62]. Compared with general robots, Zenbo's price is more reasonable, which increases its popularity for general families or nursing institutions. Therefore, this study uses Zenbo as an auxiliary research tool to deeply explore the application of companion robots and demand analysis of an elderly family home motion system.

#### 2.4. Mobile Application and Scoring System

The vigorous development of smartphones and wireless networks allows users to easily access the information they want and need through mobile technology anytime and anywhere. It has created diversified mobile application systems, changed the way people communicate, and improved social interactions. Therefore, more middle-aged and seniors are using mobile phones to take, upload, and share photos or use social software to communicate with friends and relatives. They are enjoying the convenience brought by information technology. According to the Ofcom [63] report on adults' multimedia use and attitudes, 28% of the elderly aged 75 or above used tablet computers in 2018, or an increase of 15% over 2015. In 2020, 77% of the elderly over 65 used the Internet at home. According to the same survey results, the Internet use rate of the age group 65 to 74 increased from 52% in 2011 to 83% in 2019. This means that seniors are narrowing the generation gap in technology use, and they are not only using their computers but also expanding to mobile phones and tablets [64].

In the "Demand and Impact of the Elderly Consumers" report for the middle-aged and the elderly, AT Kearney [65] pointed out that the elderly have high acceptance of mobile technology, 69% of seniors use the Internet and mobile phones at the same time, and mobile technology has brought changes and influence to the lives of seniors. A report on the use of smartphones published by InsightXplorer [66] explained the use of applications and browsers/web pages on smartphones. Regardless of the content type, the overall use rate of Apps is higher than that of browsers/web pages, and the availability of local services and specific information is higher than that of browsers. According to Dennison et al. [67], mobile phones have become an important way to spread healthy behaviors. The development of Internet mobile technology has promoted a rapid rise in fitness Apps, which are third-party application(s) of smartphones or wearable devices that can help users record fitness data, guide sports learning, and lead healthy lifestyles [68]. Thus, more and more scholars are continuously investing in research related to fitness or sports applications [69–72]. Under such consideration, people can effectively and immediately manage their own health conditions and make health management plans using mobile Apps.

Most intelligent sports systems are mainly targeted at young people, and their category is also biased toward running or cycling records and fitness teaching. This may motivate people to try to mimic observed activity patterns, resulting in more or fewer sports activities depending on the sports behavior of their peers. Products similar to this research include the golf practice system, which focuses on an evaluation of swing strength and golf trajectory [73,74]. Clubs on Strava [75] form the largest online social network in the field, connecting millions of runners, cyclists, hikers, walkers, and other active people. Franken et al. [76] found Clubs on Strava influence people's running behavior. A sports partner's excellent results will motivate another follower or friend to keep exercising [75]. There are few indoor sports intelligence systems for multi-person interactive seniors, so it is very important to discuss multi-person online sports systems suitable for seniors.

Based on the above, we find that using mobile Apps to promote health management products is very effective. With the prototype development of a home-based intelligent exercise system, this research can attract Taiwanese to pay attention to their own health, immediately know health management suggestions, and use this App to achieve the purpose of health maintenance. In order to adapt to the sports system used by seniors at home, this study takes golf croquet as an example to improve the croquet field and rule restrictions, based on the research result from "Auto scoring system for golf strike back ball" published by Chen et al. [77].

This system is part of a new patent application (No.: TWM602930). As the basis for the score of golf croquet, it uses a sensing plate and image recognition device (as shown in Figure 3) to determine the path and final stop position of the croquet after hitting. The score is recorded and displayed in the App software. The "Smart Golf Croquet System" App in Figure 4 includes five functions: login, select ball, swing ball, score, and leader board. It also can automatically sort the scores and list the rankings, so as to reduce the number of judges, scorers, and other staff. A smaller venue can also be used. This system improves the original restrictions and also records the muscle groups and fitness conditions used by users in the process of use because the App software of this system contains an image recognition function module and storage area that can identify and store at least three hitting paths. Thus, the base point of hitting can be set in advance with assistive devices. Rehabilitation therapists suggest that three hitting points be determined for fixed



muscles. The hitting path, the position of the ball, and scoring are recorded at the same time, and so specific muscle rehabilitation or fitness status can be evaluated.

Figure 3. Ultrasonic sensors plate and image recognition device.



Figure 4. An App for the auto scoring system for golf strike back ball.

When playing traditional golf croquet, there will often be an embarrassing situation where the ball is pressing on the line or many points. At this time, players will be confused about the score. Our work changes it to an intelligent scoring method, using image recognition to detect the actual score of the ball, sending it back to the background to calculate the score, and then making a sound to inform the score. This is convenient to help users easily know the score overview. The intelligent scoring system mainly uses the camera that shoots the scoreboard, backend differential image recognition, and finally, the score read-out.

In smart golf croquet, there are four main items: smart scoring system, smart club, robot, and App. The intelligent scoring system mainly uses the camera that shoots the scoreboard, then loads the differential image recognition at the backend, and finally reads the score out loud. The smart club will have installed a shock, pressure, and accelerometer sensor, and an Arduino processor will process the data. The robot has an intelligent voice and camera system. Usually, the elderly is not an ideal age group for using electronic products and can operate this system only by speaking. Furthermore, the App screen can be directly displayed on the robot's screen. The last part of the App integrates the data of previous scoring and clubs for display. Next, multiplayer connection is the most important part, so that players can play together online without being disturbed by space factors.

#### 2.5. TAM and Hypotheses

TAM, published by Davis et al. [20], is based on the theory of rational behavior. Davis believed that an individual's attitude will affect his/her willingness to use something and then affect his/her actual behavior. They proposed that "perceived usefulness" and "perceived ease of use" are the key factors that affect his/her acceptance or not and added external variables, including system design characteristics, user characteristics, use environment, task characteristics, degree of involvement, etc., to explore their relevance with a user's cognition and behavioral intention. In other words, because of the influence of some external factors, users realize whether the new technology is useful and easy to use, and only after they have a positive attitude toward the internal psychology, will they have the intention to use, and then the actual use behavior will occur.

The TAM dimensions defined by Davis et al. [20] are as follows. (1) Perceived usefulness: external variables and perceived ease of use affect perceived usefulness, improvements in the ease of using the system promote work efficiency, and perceived usefulness affects people's attitude and intention to use new technology systems. (2) Perceived ease of use: perceived ease of use affects perceived usefulness and use attitude because users need to be able to use it to produce performance. The easier the technology system is to use, the better the work efficiency will be, and the time to complete the task will be shortened. (3) Intention to use: perceived ease of use and attitude to use affect people's intention to use the technology system. We expect that the technology system can improve work performance and feelings about the system and encourage users to actually use the product. Research on the acceptability of interpreting information systems based on TAM has obtained much empirical evidence [78,79], which has been fully validated in terms of both interpretation ability and theoretical applicability. Although TAM was originally only used in information-related fields, many studies have applied it to information services and innovation and technology products [80,81], proved its effectiveness, and verified the relationships among users' perceived usefulness, perceived ease of use, use attitude, intention to use, user behavior, and other aspects of information technology.

#### 2.5.1. Mobile Phone Experience and Education Level

Experience and education level are decisive factors of intention to adopt, which directly leads to people's willingness to use home-based intelligent exercise systems. Experience indicates the benefits that customers can obtain when using home-based intelligent exercise systems, which can be a judgment of the total value of the home intelligent sports system [82]. According to one study, experience and education level significantly affect

people's attitudes toward home-based intelligent exercise systems. Schill et al. [83] pointed out that customers' experience in using home-based intelligent exercise systems determines their purchase intention. Tichenor et al. [84] noted that parents with higher education levels have higher knowledge and experience in using new scientific and technological information and a higher willingness to adopt electronic children's books. Youn and Lee [85] suggested that experience may affect perceived usefulness and perceived ease of use when using mobile payment. We thus present the following hypotheses.

**H1.** *Mobile phone experience has a significantly positive impact on the perceived usefulness of a home-based intelligent exercise system.* 

**H2.** Education level has a significantly positive impact on the perceived ease of use of a home-based intelligent exercise system.

#### 2.5.2. Perceived Usefulness and Perceived Ease of Use

People's attitudes, willingness, and actual use behavior toward the acceptance and use of science and technology are mainly influenced by their perceived usefulness and ease of use of the information system. According to Peng and Yan [86], the perceived usefulness of scientific and technological products directly affects the willingness to use multiple media kiosks (MMKs). Hsu and Chang [87] predicted students' views on the acceptance and use of Moodle, an open-source e-learning system. The results showed that perceived ease of use and perceived usefulness have a significantly positive impact on the attitude toward using Moodle. Wang et al. [88] discussed how to make artificial intelligence (AI) more effective and profitable in e-commerce and how entrepreneurs can use AI technology to help achieve their business goals. The results of model verification showed that perceived usefulness and perceived ease of use also have a positive impact on use attitude and intention. Based on the above research, we make the following hypotheses.

**H6.** *Perceived usefulness has a significantly positive impact on the intention to use a home-based intelligent exercise system.* 

**H7.** *Perceived ease of use has a significantly positive impact on the intention to use a home-based intelligent exercise system.* 

#### 2.6. HNT and Hypotheses

Maslow's hierarchy of needs theory [25] is the most widely used theory in the study of organizational incentives. It aims to reflect the general needs of society as its foundation and has been developed into more acquired emotions. According to this theory, in order to generate motivation in the next stage, each previous stage must be satisfied by the individual, and the greatest function of the hierarchy of needs theory is that it points out that everyone has needs [89]. This theory is important in explaining personality and motivation. It puts forward that the internal motivation of individual growth is motivation, motivation is composed of various needs of different levels and natures, there are high and low levels and sequences among various needs, and the degree of demand and satisfaction at each level will determine the realm of individual personality development [90].

The theory includes the following requirements. (1) Physiological needs: the basic needs for human survival, including food (hunger, thirst), clothing, housing, transportation, education, and happiness. (2) Safety needs: including physical and psychological safety and stability, avoiding physical injury or mental injury. (3) Social needs: various interpersonal relationships for personal needs, such as friendship, love, companionship, and sense of belonging. (4) Self-esteem needs: personal needs related to self-esteem, such as the pursuit of social status, social respect, recognition, and trust. (5) Cognitive needs: seeking knowledge and the curiosity of individuals to explore specific problems

10 of 32

and knowledge and then learn. (6) Aesthetic needs: seeking beauty, including the appreciation of external beauty and the internal desire to achieve perfection in the things within. (7) Self-fulfillment needs: the highest level of demand in this theory refers to the individual's ability to maximize, realize his/her ideal and ambition, and become his/her desired person. (8) Spiritual needs: cover the needs to transcend oneself and integrate nature and humans, including altruism, compassion, and other goodness implied in human nature.

Maslow's hierarchy of needs theory is not only a prominent science in psychology but also an important exposition of humanistic psychology and is widely used in other fields. Ryan et al. [91] proposed a framework for community stability and sustainability during COVID-19. Their study pointed to the need for policymakers to understand associated risks and how Maslow's hierarchy of needs and social determinants of health can guide policy across society. Aligning decision-making with societal needs will help ensure that the needs of all segments of society are met while managing a crisis. Altymurat et al. [92] discussed the application of Maslow's hierarchy of needs principles in organizations, showing that companies will function optimally when the requirements for confidentiality, convenience, and certainty are met, thereby enabling the satisfaction of the process of user knowledge requirements to go smoothly. Hale et al. [93] proposed Maslow's hierarchy of human needs (physiological, safety, love/belonging, esteem, and self-actualization) as a potential framework for addressing wellness programs. Their findings revealed that widespread burnout in graduate medical education exists and has detrimental effects on career satisfaction, personal well-being, and patient outcomes.

#### 2.6.1. Education Level

Although it is easy to meet an individual's low initial social, self-esteem, and self-realization needs, such needs may increase with education level. Orth et al. [94] studied the development of self-esteem from youth to old age. The results indicated that people with higher education have higher self-esteem than those with lower education. Rational users will try to correct this disharmony by distorting or modifying their sense of social existence, emotional belonging, self-expression, and happiness so as to make it more consistent with reality [95,96]. Van Eckert et al. [97] pointed out that the self-esteem level of nurses with high academic qualifications is significantly higher than that of nurses with low academic qualifications, and the relative job satisfaction and high-quality patient care are significant. Therefore, education level will often improve users' satisfaction with social needs, self-esteem, and self-realization needs, while negation will reduce this satisfaction. Based on the above research, we make the following hypotheses.

**H3.** Education level has a significantly positive impact on social needs.

**H4.** Education level has a significantly positive impact on esteem needs.

**H5.** Education level has a significantly positive impact on self-actualization needs.

#### 2.6.2. Social Needs

Emotional belonging refers to intimate feelings and emotional contact between individuals, including contact intensity and moral support [98]. Sinclair and Dowdy [99] noted that emotional belonging will promote individuals to perceive intimacy in the group and establish better interpersonal relationships with others. Ridings and Gefen [100] stated that the most influential motivation to join a virtual community is to seek social support and friendship. Rau et al. [101] believed that the main motivation for people to participate in social networks is to meet their emotional needs rather than information needs. Therefore, we believe that the satisfaction of emotional belonging significantly affects users' attitudes toward a home intelligent sports system and then affects their willingness to continue to use it. Based on the above research, we present the following hypothesis. **H8.** Social needs have a significantly positive impact on the intention to use a home-based intelligent exercise system.

#### 2.6.3. Esteem Needs

Cui et al. [21] confirmed that self-esteem significantly affects the purchase intention of electric vehicles. Lu et al. [102] also supported this hypothesis by investigating wireless Internet service products. In addition, Bi and Zhang [103] proposed a relatively new interactive marketing model: influencer marketing, and the research results showed that the level of self-esteem significantly affects the purchase of endorsed products. People with high status are also prone to accept environment-friendly products to enhance themselves [104]. However, most studies have only demonstrated that self-esteem may affect product adoption intention, neglecting how self-esteem impacts the adoption of intelligent products. Therefore, we propose the following hypothesis.

**H9.** Esteem needs have a significantly positive impact on the intention to use a home-based intelligent exercise system.

#### 2.6.4. Cognitive Needs

Human beings have the need to increase their intelligence and pursue knowledge. Cognitive needs (CNs) form the expression of humans' natural need for learning, exploring, discovering, and creating so as to better understand the world. If this demand for selfrealization and learning growth is not met, then it will lead to confusion and an identity crisis, which directly relates to the need for exploration or the openness of experience [25]. Lin et al. [105] used cognitive needs as a regulator to explore the influence of online comments on purchase intention and found that shoppers with high cognitive needs have a positive impact on purchase intention. Hussain and Shabir [106] pointed out that information professionals need to re-examine the use of social media to meet their cognitive needs because the satisfaction they acquire differs from the satisfaction they seek from social media. Zhao and Zhu [107] discussed the role of cognitive needs in regulating internal mechanisms, and their results showed that cognitive needs, as a regulating variable, significantly affect the influence path from perceived usefulness to attitude and purchase intention. At present, there is no research on testing home-based intelligent exercise systems with the CN variable and linking it with TAM. Thus, we put forward a hypothesis.

**H10.** Cognitive needs have a significantly positive impact on the intention to use a home-based intelligent exercise system.

#### 2.6.5. Aesthetic Needs

Aesthetics needs (ANs) are a higher level of Maslow's needs. With the development of the economy, most people can meet the basic level of Maslow's needs, and visual aesthetics (VAs) become more important. As far as a company is concerned, VAs are a symbol that distinguishes it from other competitors. The best example is Apple, which is a powerful measure for the company to compete for increasing customers [108]. According to studies [109–112], the fact that ANs significantly affect customers' willingness to use has been proved. However, at present, there is no research on testing a home intelligent sports system with the AN variable and linking it with TAM. Thus, we put forward another hypothesis.

**H11.** Aesthetic needs have a significantly positive impact on the intention to use a home-based intelligent exercise system.

### 2.6.6. Self-Actualization Needs

The satisfaction of self-realization needs includes the satisfaction of self-expression and the satisfaction of happiness. Self-expression is defined as an individual's desire to show himself/herself [113]. Daimi and Tolunay [114] researched social media influencers and their influence on consumers' purchase decisions, noting that the authenticity and credibility of influencers and the self-demand of followers significantly affect their purchase intentions. In addition, both eudaimonia and hedonia relate to life satisfaction and positive emotion [115]. Therefore, when users think that a higher level of self-expression satisfaction can enhance social attraction, they will attain psychological satisfaction, which will further enhance their sustained willingness. Based on the above research, we propose the following hypothesis.

## **H12.** Self-actualization needs have a significantly positive impact on the intention to use a homebased intelligent exercise system.

To sum up the above, this study is based on perceived usefulness and perceived ease of use in TAM to explore the relationships among users' cognition, attitude, intention, and use of technology systems. Next, we explore the needs of seniors, propose to add Maslow's needs theory to improve TAM, and confirm the relationships among social needs, selfesteem needs, cognitive needs, aesthetic needs, and self-actualization needs in Maslow's needs theory, as well as perceived usefulness, perceived ease of use, and intention to use of TAM.

#### 3. Research Method and Design

This study proposes to add Maslow's hierarchy of needs theory to improve TAM and explores the needs of seniors for healthy sports mobile phone apps. It includes the relationship between TAM of perceived usefulness, the perceived ease of use, and Maslow's needs theory of social needs, esteem needs, cognitive needs, aesthetic needs, and self-actualization needs. The control variables include the relationship between mobile phone use experience and perceived usefulness and the relationship between education level and perceived usefulness, social needs, esteem needs, and self-actualization needs. This study establishes hypotheses by exploring the relationship between these variables and the home-based intelligent exercise system of golf croquet and carries out experimental analysis using a questionnaire. The research structure is shown in Figure 5.

According to the literature, perceived usefulness, perceived ease of use, and intention to use TAM are sorted into 7 criteria and implications and 16 items. Among them, Davis and Bagozzi [20] predicted people's acceptance of computers by measuring their intentions and their ability to explain their intentions based on their attitudes, subjective norms, perceived usefulness, perceived ease of use, and related variables, thereby improving organizational performance. Venkatesh and Davis [37] developed and tested a theoretical extension of TAM, explaining the perceived usefulness and use intention of social influence and cognitive tool process. The results showed that both the social influence process and cognitive tool process significantly affect user acceptance. Huang [116] applied TAM to explore the use of a tele-healthcare system by community residents, as well as the acceptance of nursing technology products. Kuo [117] used TAM II to explore the influence of subjective norms, impressions, mission relevance, output quality, and results explicability on consumers' cognitive usefulness of electric vehicles and to explore the influence of cognitive usefulness and cognitive usability conditions on consumers' intention to use electric vehicles. The application of TAM by Tang [118] aimed to explore the actual use of digital learning by indigenous adults and their acceptance of technology.



Figure 5. Conceptual framework.

The above criteria and their implications are as follows. (1) Relieve loneliness: can effectively alleviate loneliness. (2) Stay healthy: can maintain physical and mental health. (3) Convenience: very convenient to use. (4) Easy to operate: operation steps are simple. (5) Easy to learn: does not take much time to learn. (6) Self-intention to use: facilitates users to refer to the product and use it in practice. (7) Recommend and share: recommend and share the product with others.

According to the literature, the five items in Maslow's hierarchy of needs theory are sorted out into 15 criteria and implications and 15 items. Among them, Hsu [119] aimed to understand the importance and satisfaction of middle-aged and elderly people at the level of needs and to explore the distribution of Maslow's five major needs by using the analysis of importance and performance (IPA). The research results showed middle-aged and elderly people with the ability to prepare in advance and be healthy and active in aging in the future. Chen [120] observed the phenomenon and significance of rural religious belief behavior from tradition to the present with Maslow's demand theory and discussed the feasibility of developing rural religious creation and activities with the concept of leisure. That study also put forward suggestions for practical development according to the research results.

The above criteria and their implications are as follows. (1) Sense of identity: can identify with each other. (2) Maintain good relationships: can maintain good relationships with others and maintain emotions. (3) Embrace other people: able to accept others and live in harmony. (4) Sense of glory: can exhibit a sense of pride. (5) Earn recognition: able to gain the affirmation of others. (6) Respected by others: be respected by others. (7) Full of confidence: can make oneself full of confidence. (8) Curious and seek knowledge: be curious about new sports equipment and want to know. (9) Learning their skills: easy to learn and want to learn skills for use. (10) Hone their skills: want to use technology frequently to improve skills. (11) System functions: functional design meets the requirements. (12) Font display: the font and screen display are clear and easy to recognize. (13) Exert personal potential: be able to fully exert one's potential without being disturbed by emotion or environment. (14) Competition with myself: actively face challenges and do not give up easily. (15) Peak experience: experience a high degree of self-realization and feel happy, excited, and touched in a short period of time, even selfless feelings.

All items are measured using the Likert five-point scale, which is divided into five options according to the degree of agreement: very disagree (1 point), disagree (2 points), ordinary (3 points), agree (4 points), and very agree (5 points), as shown in Table 1. In this paper, the selected questionnaire items in Appendix A.

Dimension	Criterion	Implication	References
Perceived usefulness (PU)	Relieve loneliness Stay healthy	Can effectively alleviate loneliness. Can maintain physical and mental health.	[21 40]
Perceived ease of use (PE)	Convenience Easy to operate	Very convenient to use. Operation steps are simple.	[31,49]
	Easy to learn	Does not take much time to learn.	
Intention to use (IU)	Self-intention to use Recommend and share	Facilitate users to refer to the product and use it in practice. Recommend and share the product with others.	[106–108]
	Sense of identity	Can identify with each other.	
Social needs (SNs)	Maintain good relationships	Can maintain good relationships with others and maintain emotions	
	Embrace other people	Able to accept others and live in harmony.	
Esteem needs (ENs)	Sense of glory Earn recognition Respected by others Full of confidence	Can produce a sense of pride. Be able to gain the affirmation of others. Be respected by others. Can make oneself full of confidence.	
Cognitive needs (CNs)	Curious and seek knowledge Learning their skills Hone their skills	Be curious about new sports equipment and want to know. Easy to learn and want to learn skills in use. Want to use more to improve skills.	[109,110]
Aesthetic needs (ANs)	System functions Font display	Functional design meets the requirements. Font and screen display are clear and easy to recognize.	
Self-actualization needs (SA)	Exert personal potential Competition with myself Peak experience	Be able to fully exert their potential without being disturbed by emotion or environment. Actively face challenges and do not give up easily. Experience a high degree of self-realization, and can feel happy, excited, and touched in a short period of time, even selfless feelings.	

Table 1. Literature compilation of dimension.

During the literature review, 31 questions in total were collected from a questionnaire consisting of 7 criteria for TAM and 15 criteria for Maslow's hierarchy of needs theory. First, a sample structure analysis is conducted on the collected questionnaire, and narrative statistics are analyzed and explained for each aspect. Second, confirmatory factor analysis (CFA) is conducted to confirm the results after clustering. CFA is a statistical method used in social sciences to evaluate the degree of fit between theoretical models and real-life collected datasets. It is often used as a construct validity test scale or measurement tool. In other words, its purpose is to test the relationship between observed indicators and potential variables (factors) in the model [121]. Third, regression analysis is conducted to validate the proposed hypothesis. Regression analysis is the most basic and important statistical analysis technique and hypothesis validation method in social science research methods, which is used to display the relationship between two or more variables [122,123]. Finally, an analysis of variance is conducted on the dimensions proposed in this study based on education level and mobile phone usage experience. Analysis of variance helps identify whether differences between data groups are statistically significant. Its principle is to analyze the level of differences within each group by selecting samples from each group [124].

This is a non-interventional study (e.g., surveys, questionnaires, social media research). The process was anonymous and all participants were fully informed as to why the study was being conducted, how the data would be used, and whether there were any associated risks. Based on the regulations announced in Taiwan: Human Subjects Research Act, Chapter 2, Article 5, Announcement of Ministry of Health and Welfare, Executive Yuan, Republic of China (Taiwan) Issue Date: 5 July 2012, Issue No.: 1010265075, this study was conducted

under the conditions of 'Exempt from being submitted to the Ethics Review Committee for review', please refer to Appendix B for detailed regulations and ethical statements.

This study collected empirical data on subjects from two elderly care centers in northern Taiwan using a cross-sectional survey and purposive sampling. The main subjects are seniors over 65 years old and their accompanying caregivers. The unit under test is a long-term partner of this study. Every semester, students act as volunteers to serve and visit the elderly residents, providing timely physical and mental comfort and support to them, bringing adjustments, leisure, and entertainment to their lives, and thereby improving their quality of life. The quality of life of the residents can be improved by providing timely relief and comfort for the residents' illnesses and leveraging the power of social resources to achieve a community care model and social interaction.

In order to carry out this research smoothly, before the test, a discussion was held with the person in charge of the two elderly care centers about the relevant content, including the location of the meeting room, space capacity, related equipment, the location of golf croquet, and the size of the venue. The person in charge was asked to communicate to the elderly and accompanying caregivers who can participate in this activity and then compile the number of people willing to participate. In order to carry out this activity smoothly, in addition to stipulating that elderly people who were willing to participate needed to be accompanied by a companion or caregiver who can assist them, in order to maintain a safe distance and avoid contracting COVID-19, the number of people per day/session was limited to 32 people (16 groups, 2 people in each group). The relevant number of participants and the test schedule are shown in Table 2. The total number of participants in elderly care center (A) is 192, and the total number of participants in elderly care center (B) is 128. Due to the large number of people in elderly care center (A), the testing dates were held every Tuesday and Thursday for three weeks, while those for elderly care center (B) were scheduled every Wednesday for four weeks.

Institution Week Time	Tuesday/Thursday					
Institution (week   Inne	13:00-14:00	14:00-15:30	15:30-16:30	Participants		
Elderly care center (A)	Explanation/video introduction	Experience golf croquet	Fill out the questionnaires	192		
Institution\Week\Time —	Wednesday					
	13:00-14:00	14:00-15:30	15:30–16:30	Participants		
Elderly care center (B)	Explanation/video introduction	Experience golf croquet	Fill out the questionnaires	128		

Table 2. Implementation schedule.

Elderly care center (A): Every Tuesday and Thursday for a total of three weeks. Elderly care center (B): Every Wednesday for four weeks. The number of people tested every day is 32 (16 groups, 2 people in each group).

Our work uses a physical questionnaire. Before filling out the questionnaire, the individual seniors and their companion or caregivers gathered in a conference room and were verbally explained the rules and disadvantages of traditional golf croquet. The improvement methods proposed in this study were introduced. At the same time, recorded explanations and introduction videos were played to make the participants more aware of the overall testing procedure. After completing the explanation, they were moved to a spacious indoor space to experience the golf croquet home-based intelligent exercise system. Finally, they gradually filled out the questionnaire content based on their experience results. For those who did not understand the questionnaire content, we provided assistance and explained the content of the questions.

The greater the number of samples that are selected, the higher is the accuracy of inferring the actual situation of the statistical analysis results, which can prevent differences from being effectively tested due to too few samples and can reduce the probability of type II errors. However, when the number of samples is too large, subtle differences may cause

the research results to reach significant results, increasing the chance of type I errors [125]. The number of samples selected should be adjusted according to the needs of the statistical analysis used in this study. It is best to have more than 300 samples for factor analysis [126], or the number of effective samples should be at least 5 times the number of microscale questions [127]. In addition, depending on the number of questions in the questionnaire, the sample size in this study should be 3 to 5 times or 5 to 10 times the number of questions in the largest subscale [128,129]. The number of questionnaires is 320. The questionnaires were completely recovered, and so the effective recovery rate is 100%, thus presenting sufficient sample representativeness. The following sections describe the data collection and analysis of the results.

#### 4. Data Collection and Analysis of Results

#### 4.1. Sample Description

The basic data of the collected questionnaire samples are as follows: 82.2% of the respondents are male, 78.4% of the respondents have a high school education, and 80.6% have more than five years of experience in using smartphones. Among the subjects, 31 have chronic diseases, with cardiovascular diseases accounting for the majority (52%), followed by diabetes (26%), and 61% have had chronic diseases for more than five years. Those who participate in sports three times a week account for 49.7% of the sample, followed by more than five times a week at 26.3%. A duration of each exercise of less than 30 min accounts for the majority (74.4%), followed by 1 h (19.4%). The majority (37.2%) of them have continued to exercise for more than five years. The basic information collected with the questionnaire is shown in Table 3.

Item	Information	Quantity	Percent
	Male	263	82.2
Gender	Female	57	17.8
	Elementary school	3	0.9
Education loval (EL)	Junior high school	20	6.3
Education level (EL)	High school	251	78.4
	University or above	46	14.4
	None	4	1.3
	Less than 1 year	0	0
Mobile phone experience (ME)	1~3 year(s)	9	2.8
	3~5 years	49	15.3
	5 years or above	258	80.6
	None	289	90.3
Chronic disease	Yes (including cardiovascular disease, diabetes, rehabilitation	21	0.7
	treatment, hypertension, cholesterol, and others)	51	9.7
	None	289	90.3
	Less than 1 year	3	0.9
Seniority of chronic disease	1~3 year(s)	8	2.5
	3~5 years	1	0.3
	5 years or above	19	5.9
	0	15	4.7
	1	11	3.4
	2	39	12.2
Exercise times/week	3	159	49.7
	4	12	3.8
	5 or above	84	26.3

Table 3. Basic information of respondents.

Item	Information	Quantity	Percent
	Less than 30 min	238	74.4
	1 h	62	19.4
Exercise duration/time	1~2 h(s)	15	4.7
	2~3 h	4	1.3
	3 h or above	1	0.3
	Less than half year	25	7.8
	Less than year	32	10.0
Seniority of sport	1~3 year(s)	62	19.4
, <u>1</u>	3~5 years	82	25.6
	5 years or above	119	37.2

Table 3. Cont.

According to the statistical results in Table 4, the average number of questions on perceived usefulness, perceived ease of use, intention to use, social needs, self-esteem needs, cognitive needs, aesthetic needs, and self-actualization needs of the respondents ranges from 3.97 to 4.13. This represents the respondents' positive attitude toward the home action system of golf croquet.

Table 4. Descriptive statistics of question.

Dimension	Criterion	Question No.	Mean	SD
		PU1	4.10	0.494
	Relieve loneliness	PU2	4.00	3.92
Perceived usefulness (PU)		PU3	3.97	0.461
· · · · · · · · · · · · · · · · · · ·	Staying healthy	PU4	4.09	0.401
		PU5	4.04	0.402
	Contraction	PE1	4.20	0.461
	Convenience	PE2	4.19	0.446
Perceived area of use (PE)	Fasy to operate	PE3	4.11	0.411
referved ease of use (FE)	Lasy to operate	PE4	4.13	0.424
	Fasy to learn	PE5	4.11	0.431
	Lasy to learn	PE6	4.10	0.410
		IU1	4.05	0.456
	Self-intention to use	IU2	4.03	0.429
Intention to use (IU)		IU3	4.10	0.447
	December 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	IU4	4.01	0.407
	Recommend and share	IU5	4.01	0.375
	Sense of identity	SN1	3.99	0.392
Social needs (SNs)	Maintain good relationships	SN2	4.07	0.424
	Embrace other people	SN3	3.99	0.363
	Sense of glory	EN1	3.95	0.401
Esteen needs (ENs)	Earning recognition	EN2	3.97	0.375
Esteent needs (EINS)	Respected by others	EN3	3.97	0.414
	Full of confidence	EN4	4.02	0.461
	Curious and seek knowledge	CN1	4.03	0.410
Cognitive needs (CNs)	Learning their skills	CN2	4.00	0.423
	Hone their skills	CN3	4.02	0.419
	System functions	AN1	4.03	0.429
Aesthetic needs (AINS)	Font display	AN2	4.03	0.399
	Exert personal potential	SA1	3.99	0.434
Self-actualization needs (SA)	Competition with myself	SA2	3.98	0.415
	Peak experience	SA3	4.00	0.419

#### 4.2. Factor Analysis, Reliability, and Validity Analysis

To further understand the degree of consent of the subjects to each dimension, this study carries out factor analysis, respectively, for TAM's perceived usefulness, perceived ease of use, and intention to use as well as Maslow's social needs, self-esteem needs, cognitive needs, aesthetic needs, and self-actualization needs. It extracts common factors with a characteristic value greater than 1 [130] using principal component analysis and the maximum variation pivot method. Questions with an absolute value of the factor load greater than 0.5 are reserved and properly classified. In terms of scale reliability, according to Guilford [131], Cronbach's  $\alpha$  greater than 0.7 means high reliability and a value less than 0.35 shows low reliability, which should be rejected [132]. The construction validity of this study is based on the item-total correlation method of Kerlinger [133]—that is, assuming that the total score is valid, the size of the correlation coefficient between individual items and the total score is the measure of construct validity.

The results in Table 5 show that the Kaiser–Meyer–Olkin (KMO) value of each facet is between 0.500 and 0.851. KMO is an index used to compare simple correlation coefficients and partial correlation coefficients between variables. The closer its value is to 1, the stronger the correlation between variables, and the more suitable the original variables are for factor analysis. On the contrary, the closer it is to 0, the weaker the correlation between variables, and the less suitable it is [134,135]. Bartlett's ball test is significantly in line with the requirements, and the values of each facet are quite good.

	Support	КМО	Bart	<b>Bartlett's Test of Sphericity</b>	
Surface		$\chi^2$	df	p	
P	U	0.812	511.938	10	0.000 ***
F	Έ	0.851	2158.924	15	0.000 ***
Ι	U	0.847	1029.735	10	0.000 ***
S	N	0.722	527.225	3	0.000 ***
E	N	0.789	1006.788	6	0.000 ***
С	N	0.761	714.099	3	0.000 ***
A	N	0.500	258.196	1	0.000 ***
S	А	0.729	567.436	3	0.000 ***
S E C A S	N N N A	0.722 0.789 0.761 0.500 0.729	527.225 1006.788 714.099 258.196 567.436	3 6 3 1 3	0.000 *** 0.000 *** 0.000 *** 0.000 ***

Table 5. KMO measure of sampling.

PU: perceived usefulness. PE: perceived ease of use. IU: intention to use. SN: social needs. ENs: esteem needs. CN: cognitive needs. AN: aesthetic needs. SA: self-actualization needs. \*\*\* p < 0.001.

The results in Table 6 show that the factor load of each item is between 0.502 and 0.877, which is more than 0.5. The characteristic value of each facet is between 1.746 and 4.711, the cumulative explanatory variance is between 57.773 and 87.302, and the item-total correlation value of each item is greater than 0.5, indicating considerable constructive validity and content validity. Cronbach's  $\alpha$  values are all greater than 0.6, denoting good internal reliability. There is thus real correlation between the measurement items, and the content of the questionnaire is highly consistent.

#### 4.3. Hypothesis Testing

Table 7 shows that, to predict the intention of use, each dimension uses SPSS software to carry out regression analysis. According to the research framework (Figure 5), it is assumed that the path direction variable is the dependent variable, and the path starting variable is the independent variable or predictive variable. According to the analysis of the relationship between perceived usefulness and intention to use, the empirical results show that hypothesis H6 is supported, indicating that perceived usefulness has a significantly positive impact (p < 0.001) on intention to use.

Question No.	Eigenvalue	Cumulative Percentage	Factor Loading	Cronbach's α	Item-Total Correlation
PU1			0.622		0.770
PU2			0.660		0.758
PU3	2.889	57.773	0.556	0.814	0.783
PU4			0.548		0.782
PU5			0.502		0.793
PE1			0.730		0.939
PE2			0.777		0.934
PE3	4 771 1	79 510	0.777	0.045	0.935
PE4	4./11	76.312	0.797	0.945	0.933
PE5			0.828		0.931
PE6			0.803		0.933
IU1			0.692		0.885
IU2			0.638		0.894
IU3	3.611	72.214	0.761	0.902	0.871
IU4			0.792		0.869
IU5			0.728		0.881
SN1			0.841		0.788
SN2	2.413	80.425	0.735	0.874	0.883
SN3			0.836		0.797
EN1			0.755		0.907
EN2	2 220	90 742	0.871	0.017	0.873
EN3	3.230	80.742	0.840	0.917	0.879
EN4			0.764		0.909
CN1			0.852		0.898
CN2	2.595	86.512	0.877	0.922	0.877
CN3			0.866		0.866
AN1	1 746	87 202	0.873	0.853	0.880
AN2	1./40	87.302	0.873	0.000	0.862
SA1			0.867		0.801
SA2	2.643	82.09	0.787	0.891	0.874
SA3			0.809		0.856

Table 6. Analysis of factor, reliability, and item-total correlation.

Question Nos. refer to Appendix A.

 Table 7. Regression analysis results.

Dimension	β	Т	p	$R^2$	$Adj-R^2$
PU	0.700	17.464	0.000 ***	0.490	0.488
PE	0.615	13.909	0.000 ***	0.378	0.376
SN	0.793	23.174	0.000 ***	0.628	0.627
EN	0.779	22.139	0.000 ***	0.607	0.605
CN	0.770	21.545	0.000 ***	0.593	0.592
AN	0.798	17.854	0.000 ***	0.501	0.499
SA	0.742	19.738	0.000 ***	0.551	0.549

PU: perceived usefulness. PE: perceived ease of use. SN: social needs. EN: esteem needs. CN: cognitive needs. ANs: aesthetic needs. SA: self-actualization needs. \*\*\* p < 0.001.

According to the analysis of the relationship between perceived usefulness and intention to use, the empirical results show that hypothesis H7 is supported. This indicates that perceived usefulness has a significantly positive impact (p < 0.001) on intention to use. According to the analysis of the relationship between social needs and intention to use, the empirical results show that hypothesis H8 is supported. This indicates that social needs have a significantly positive impact (p < 0.001) on intention to use.

According to the analysis of the relationship between esteem needs and intention to use, the empirical results show that hypothesis H9 is supported. This indicates that esteem needs have a significantly positive impact (p < 0.001) on intention to use.

According to the analysis of the relationship between cognitive needs and intention to use, the empirical results show that hypothesis H10 is supported. This indicates that cognitive needs have a significantly positive impact (p < 0.001) on intention to use.

According to the analysis of the relationship between aesthetic needs and intention to use, the empirical results show that hypothesis H11 is supported. This indicates that aesthetic needs have a significantly positive impact (p < 0.001) on intention to use.

According to the analysis of the relationship between the need for self-actualization and the intention to use, the empirical results show that hypothesis H12 is supported. This indicates that the need for self-actualization has a significantly positive impact (p < 0.001) on the intention to use.

Table 8 shows the difference between education level and mobile phone experience in terms of perceived usefulness, social needs, esteem needs, and self-actualization needs when analyzed using SPSS software. According to the research framework (Figure 5), the variable of path direction is assumed to be the dependent variable, and the variable of path starting is assumed to be a factor.

Between Group	SS	MS	F	p
ME-PU	4.131	1.377	1.382	0.248
EL-PU	0.726	0.242	0.240	0.868
EL-SN	6.992	2.331	5.900	0.071
EL-EN	16.921	5.640	4.015	0.001 **
EL-SA	11.713	3.904	4.015	0.008 **

Table 8. Analysis of variance, ANOVA.

ME: mobile phone experience. PU: perceived usefulness. EL: education level. SN: social needs. ENs: esteem needs. SA: self-actualization needs. SS: sum of squares of deviation from mean. MS: mean-square. \*\* p < 0.01.

According to the analysis of the relationship between mobile phone experience and perceived usefulness, the actual results of this study show that hypothesis H1 is not supported. This indicates that the positive impact of mobile phone experience on perceived usefulness is not significant.

According to the analysis of the relationship between education level and perceived usefulness, the actual results of this study show that hypothesis H2 is not supported. This means that the positive impact of education level on perceived usefulness is not significant.

According to the analysis of the relationship between education level and social needs, the actual results of this study show that hypothesis H3 is not supported. This indicates that the positive impact of education level on social needs is not significant.

According to the analysis of the relationship between education level and esteem needs, the actual results of this study show that hypothesis H4 is supported. This indicates that education level has a significantly positive impact (p < 0.01) on esteem needs.

According to the analysis of the relationship between education level and selfactualization needs, the actual results of this study show that hypothesis H5 is supported. This means that education level has a significantly positive impact on (p < 0.01) selfactualization needs.

#### 4.4. Findings and Discussion

Based on TAM and HNT, this study constructs a demand model to confirm the needs of seniors for a healthy home-based intelligent exercise system. After analyzing the empirical

data of 320 seniors using regression and difference analysis methods, this study finds that perceived usefulness (PU), perceived ease of use (PE), social needs (SNs), esteem needs (ENs), cognitive needs (CNs), aesthetic needs (ANs), and self-actualization needs (SA) positively affect the willingness of the subjects to use the home-based intelligent exercise system of golf croquet. Based on the comprehensive analysis results, the following important findings are summarized.

# 4.4.1. The Subjects Mostly Agree with the Ease of Use of the Home-Based Intelligent Exercise System of Golf Croquet

This study indicates that the subjects are highly receptive to the home-based intelligent exercise system of golf croquet, especially in terms of perceived ease of use. The average number of each item is the highest, indicating that they believe that the system is convenient to use and that they can operate the system skillfully without the help of others.

#### 4.4.2. Aspects That Affect Intention to Use

Perceived usefulness (PU): This study notes that if a user believes that a system can increase his/her work performance or be of practical benefit to him/her, then the higher his/her willingness to use the system will be. On the contrary, the less the system helps him/her, the lower is his/her willingness to use it. The subjects recognize that the home-based intelligent exercise system of golf croquet is helpful to improve their quality of life. This will make the subjects more likely to use this system. In other words, this method has a positive effect on the limbs and waist, can strengthen muscles, and achieve exercise effects. The research findings are of great significance to the acceptance of the initial subjects. Some studies have put forward a relatively pessimistic view of the ability to predict user behavior based on subjective measurements [136,137]. The results of this study show that subjects can have a good impression of cognition when they actually communicate with each other in a group interactive way. Therefore, it is critical to ensure that the prototype of the design is fully implemented because at the initial stage, the testers expect to provide valuable insights on the acceptability of subsequently revised software/hardware products [138].

Perceived ease of use (PE): This study finds that if a user believes that a system is easy to learn, easy to use, and can operate skillfully, then the user's willingness to use the system will be higher. On the contrary, the more complex and difficult the operation process of the system is to learn, the lower the willingness to use it. The subjects think that the home-based intelligent exercise system of golf croquet is very convenient to use without any assistance from others and can be easily used skillfully, which will make the subjects likely to use the system. In other words, this study combines robot and IoT devices, which tend to be easier to use. From the perspective of knowledge and learning [139], cognitive ease of use is based on procedural knowledge. Anderson [140] suggested that procedural learning only occurs when performing skills, such as learning by doing. This is one of the reasons why procedural learning is more gradual than declarative learning. Therefore, it reflects the ease of use related to the use of technology, which requires personal experience.

Social needs (SNs): This study presents that if a user believes that a system can increase interaction with others, then his/her willingness to use the system will be higher; otherwise, the willingness to use the system will be lower. The subjects believe that the home-based intelligent exercise system of golf croquet can increase the interaction between them and their families and help them interact with each other and accept others, which will make the subjects likely to use the system. In other words, in the mode of multi-person interaction, social relations can be enhanced. Fang [141] explored the role of interaction strategies in consumer decision-making. That study pointed out that among the diverse online communication mechanisms, some customers are hesitant to shop online given that IoT cannot provide the opportunity to inspect products before purchasing, thus increasing online interactivity of the website and the addition of product information to supplement online decision-making, which increases purchase and usage intentions. Yim et al. [142] stated

that AR positively affects media and purchase intentions by generating greater novelty, immersion, interactivity, and usefulness compared with web-based product presentations.

Esteem needs (ENs): This study finds that if a system allows the user to gain confidence and self-affirmation in the process of using the system, then his/her willingness to use the system will be higher; otherwise, the willingness to use the system will be lower. The subjects believe that using the golf croquet home-based intelligent exercise system can make them feel proud, positive, confident, and respected, which will make them likely to use the system. The scoring system of golf croquet designed in this study and its ranking function can boost morale and make people feel honored. Liao et al. [143] applied the self-affirmation theory in order to examine the influence of real-world need satisfaction on online players' loyalty. Their research results showed that users' achievements and relationships in the real world can enhance their satisfaction with real-world needs, thereby enhancing the self-worth and loyalty of game players and further maintaining or enhancing the willingness to use.

Cognitive needs (CNs): This study notes that if users think that a system can arouse their curiosity and satisfy their thirst for knowledge during use, or if they want to enhance their use skills, then their perception of the system will increase, and the willingness to use will be higher. On the contrary, if the system cannot arouse the user's curiosity, then the willingness to use will be lower. The subjects think that the home-based intelligent exercise system of golf croquet can make people feel curious and want to try to use it and that learning the use skills of this system is sufficient for cognitive needs. They will thus want to hone and enhance their proficiency, which will make the subjects likely to use this system. In other words, emerging technologies such as AIoT and Apps can stimulate users' cognitive curiosity. The concept of curiosity is derived from flow theory. When people are in a state of immersion, they may be willing to interact with their environment [144]. Yoon et al. [145] investigated the influence of hedonic and utilitarian shopping values on continuous aspects that affect the intention to use online cross-border shopping. Their research results showed that hedonic value affects the continuous intention to use online shopping through the mediation of curiosity and self-efficacy.

Aesthetic needs (ANs): This study indicates that when the user thinks that a system is more recognizable in appearance, screen, and function, the higher his/her willingness to use the system will be. If the subject believes that the golf croquet home-based intelligent exercise system can be clearly identified and easy to use on the font display, and the functional design is not too complex, then the subject will likely use this system. The scoring system of golf croquet designed in this study has an elegant user interface design, which can increase the overall aesthetic feeling and thus increase the willingness to use it. Tsai et al. [146] investigated how user interface design affects the intention and attitude of the elderly to use social networking sites. Their results showed that user interface design and perceived ease of use positively relate to perceived usefulness, and an appropriate interface design will further affect adoption intention.

Self-actualization needs (SA): This study finds that if a system can make users feel a sense of achievement, excitement, and happiness in the process, then the more willing they will be to use the system. Each time the subject uses the golf croquet home-based intelligent exercise system, it is like a new challenge, which makes people want to obtain higher scores. The user process can fully exert a user's potential, gain a sense of achievement, and feel excited and happy, which makes the subject likely to use this system. In other words, learning to use mobile phone applications can achieve self-growth for users, and they are never too old to learn. Using a survey of the basic factors of students' perceived sense of achievement, pleasure, and willingness to learn web development, Zhang and Dang [147] found that the characteristics of teachers and teaching methods significantly affect their perceived sense of achievement, pleasure, and then their intention to learn web development.

#### 4.4.3. Dimensions Affected by Different Levels of Education

This study finds that people with different levels of education will have different views on esteem needs and self-actualization needs. The subjects with different education levels have significantly different views on the use of the home-based intelligent exercise system of golf croquet to gain pride, affirmation, confidence, and respect. Moreover, the subjects with different education levels have significantly different views on using the home-based intelligent exercise system of golf croquet to make people want to achieve higher scores and fully exert their potential, have a sense of achievement, and feel excited and happy during the use process. Maslow [25] referred to so-called self-actualized people, who are satisfied with life, can reach their potential and have creativity, and can have a loving and accepting attitude toward themselves and others. So-called esteem needs refer to all the needs required to acquire and maintain personal esteem, including the respect of others and self-respect. In a study on the level of education and job satisfaction, Solomon et al. [148] pointed out that it is theoretically inferred that education level involves a significant tradeoff relationship. Because of the need for self-actualization, well-educated people will enjoy more resources from the job (including income, job autonomy, and diversity). Yu and Chang [149] explored the needs of the elderly according to the type of community. They found that seniors with high education levels have better self-confidence and value of themselves as well as their level of respect and self-actualization needs.

#### 5. Conclusions

Due to economic development and the improvement in national living standards in recent years, people's leisure and sports time has increased, their concept of physical activities has become stronger, and the time and money spent on sports have risen (Malm et al. [150]). However, the aging trend in the global population highlights the planning direction of the future leisure sports of seniors. How to choose or participate in leisure sports is thus worthy of concern and discussion. Generally speaking, old people often pursue the best quality of life, but not everyone can attain or improve to a high-quality enjoyment of life when they are old. Therefore, in the future, everyone should pay attention to the planning of lifestyle in old age. The indispensable factors are life satisfaction, selfesteem, general health and function, and social status [151,152], which can also bring proper esteem and responsibility during one's own later years.

This study was conceived at the early stage of the COVID-19 epidemic. At that time, the problem of seniors living in elderly care centers in northern Taiwan for a long time and the lack of exercise and good social activities led to organ degradation or increased isolation and loneliness. Therefore, this research puts forward the concept of a home-based intelligent exercise system (including an intelligent scoring system, modified golf croquet, robot, and mobile phone applications) and applied for a patent design with the Republic of China at the same time. In order to understand the acceptance and demand level of seniors in this home-based intelligent exercise system, this study conducts demand analysis using TAM combined with Maslow's needs theory so that the original leisure sports items belonging to teams and individuals can achieve the effect of daily sports, and so that seniors can also gain psychological attribution.

#### 6. Contribution and Applicability of the Study

The results of this study show that the subjects have high acceptance of the home-based intelligent exercise system of golf croquet, especially in terms of perceived ease of use. This shows that they think the system is convenient to use and can operate the system skillfully without assistance from others. It is also a leisure activity with fitness and social functions, which is quite suitable for seniors. By participating in croquet, they can gain recognition from their peers and also temporarily put aside any unhappiness, depression, and anxiety, thus effectively finding a form of relief. These results are particularly interesting, as Chinese society has traditionally chosen to avoid unfamiliar information technology due to self-learning resistance or face-saving issues. Therefore, it is important to confirm the acceptance

level of convenience brought by emerging technologies among Chinese people. This design framework is not only used in the golf croquet system but also widely used in the seniors' sports management system. The application field can also be extended to the community, gym, or family, and the objects of use can also be extended to children and teenagers. Thus, these findings uniquely contribute to the intelligent exercise system literature.

Sports record Apps and fitness teaching Apps have both attracted researchers' attention [153,154]. Current sports and fitness Apps have many problems, such as serious product homogeneity, low user viscosity, and lack of scientific innovation. However, with continuous improvement at the professional and technical level by fitness App makers, future fitness Apps should be more convenient to meet the needs of users. In addition, with optimization and improvement of user experiences using IoT, VR, or other information technologies, Apps will become more personalized for users and take into account scientific data collection and security service provision. With the cooperation and assistance of various aspects, it is believed that sports Apps can add a healthy, safe, and happy lifestyle for older generations.

#### 7. Limitations

As with other works, this study also has certain limitations. We strived to be objective in data collection, dimensions, and criteria establishment, but due to the influence of external factors, there are still some unavoidable drawbacks. For example, subjective well-being is an individual's evaluation of the subjective perception of certain things. The subjective well-being of elderly people participating in golf croquet may have other factors that affect their subjective perception and feelings. In the future, variables such as personality traits of the elderly people themselves, life satisfaction, or other factors that may affect them can be added for further research and comparison. As this study was affected by the COVID-19 epidemic, more competition activity data could not be collected due to the inability to hold more competitions, meaning any improvement in the overall design could be slightly inadequate. It was also impossible to explain the equipment and functions of this study face-to-face with the respondents, the interaction could only maintain a relative distance, and the entire process of the activity was introduced over a video. Most of the questionnaire content in this study may have been difficult or unclear for seniors to understand. In addition, most of the seniors speak Taiwanese and do not know Mandarin Chinese well, which made communication difficult and could have led to potential misunderstandings.

**Author Contributions:** C.-L.C., S.-J.C. and B.-S.W. contributed to the conception and design of this study. Y.-L.Z. organized the database. C.-L.C. and S.-J.C. performed the statistical analysis. S.-J.C. wrote the first draft of the manuscript. Y.-L.Z., C.-L.C. and B.-S.W. wrote sections of the manuscript. All authors have read and agreed to the published version of the manuscript.

**Funding:** The authors received no financial support for the research, authorship, and/or publication of this article.

**Institutional Review Board Statement:** An institutional review board statement is not applicable to this study, which involves humans or animals.

Informed Consent Statement: Informed consent was obtained from all subjects involved in this study.

**Data Availability Statement:** The data that support the findings of this study are available from the corresponding author, S.-J.C., upon reasonable request.

**Acknowledgments:** The authors would like to thank the editors and anonymous reviewers for their time on the publication of this paper as well as the elderly participants for their contribution to the study.

**Conflicts of Interest:** The authors declare no potential conflict of interest with respect to the research, authorship, and/or publication of this article.

Appendix A					
Dimension	Criterion	Question No.	Questionnaire		
		PU1	I think using the home-based exercise system of golf croquet can help me not feel lonely.		
	Relieve loneliness	PU2	I think that using the home-based exercise system of golf croquet can increase my interaction with others.		
Perceived usefulness (PU)		PU3	I think using the home-based exercise system of golf croquet can effectively achieve the effect of daily exercise.		
	Staying healthy	PU4	I think using the home-based exercise system of golf croquet can make me think about how to obtain higher scores and train brain activity.		
		PU5	I think using the home-based exercise system of golf croquet can help me to properly exercise my muscles and joints and relieve the discomfort caused by chronic diseases.		
Perceived ease of use (PE)		PE1	I think the home-based exercise system of golf croquet does not need a large venue, which makes it very convenient for me to use.		
	Convenience	PE2	I think the home-based exercise system of golf croquet can be used without the help of many people, which makes it very convenient for me to use.		
	Easy to operate	PE3	I think it is easy for me to understand the content and function of the home-based exercise system of golf croquet.		
		PE4	I think the home-based exercise system of golf croquet can make it easy for me to use it alone.		
		PE5	I think the home-based exercise system of golf croquet can make me learn to operate it easily.		
		PE6	I think the home-based exercise system of golf croquet can make me easily and skillfully operate it.		
		IU1	I would like to use the home-based exercise system of golf croquet.		
Intention to use (IU)	Self-intention to use	IU2	I am willing to use the home-based exercise system of golf croquet with my relatives and friends.		
		IU3	If relatives and friends recommend the home-based exercise system of golf croquet to me, I will want to use it.		
	Recommend and share	IU4	I will invite relatives and friends to use the home-based exercise system of golf croquet.		
	Recommente and share –	IU5	I will share the home-based exercise system of golf croquet with my relatives and friends.		

Dimension	Criterion	Question No.	Questionnaire
	Sense of identity	SN1	I think that the home-based exercise system of golf croquet can help me gain recognition and interact with others.
Social needs (SNs)	Maintain good relationships	SN2	I think the home-based exercise system of golf croquet can make me enjoy and interact with my family.
	Embrace other people	SN3	I think the process of using the home-based exercise system of golf croquet can help me accept others and win the trust of my friends.
	Sense of glory	EN1	I think that using the home-based exercise system of golf croquet gives me a sense of honor.
Esteem needs (ENs)	Earning recognition	EN2	I think the home-based exercise system of golf croquet can make me affirmed.
	Respected by others	EN3	I think using the home-based exercise system of golf croquet can make me feel respected.
	Full of confidence	EN4	I think using the home-based exercise system of golf croquet can make me full of confidence.
	Curious and seek knowledge	CN1	I think the home-based exercise system of golf croquet can make me feel novel and want to use it.
Cognitive needs (CNs)	Learning their skills	CN2	I think learning the skills of using the home-based exercise system of golf croquet can satisfy my thirst for knowledge.
	Hone their skills	CN3	I think continuous use of the home-based exercise system of golf croquet can help me increase my skills.
A substitution of (ANIs)	System functions	AN1	I think the function design of the home-based exercise system of golf croquet can make me willing to use it.
Aesthetic needs (ANs)	Font display	AN2	I think the home-based exercise system of golf croquet can help me use the font and screen display.
Self-actualization needs (SA)	Exert personal potential	SA1	I think using the home-based exercise system of golf croquet can help me fully exert my potential and have a sense of achievement.
	Competition with myself	SA2	I think every time I use the home-based exercise system of golf croquet, it is like a new challenge, which can make me want to a higher scores.
	Peak experience	SA3	I think using the home-based exercise system of golf croquet can make me feel excited and happy.

# Appendix B

#### Ethical Statement

According to journal policy, ethical approval or exemption from the ethics committee is required before research, or authors are encouraged to cite local or national legislation that indicates that such research does not require ethical approval. This is a non-interventional study (e.g., surveys, questionnaires, social media research) where anonymity is guaranteed, and all participants are fully informed as to why the study is being conducted, how the data will be used, and whether there are any associated risks. This study does not need ethical approval according to the following Taiwan regulations: 1. Human Subjects Research Act, Chapter 2, Article 5. (https://law.moj.gov.tw/ENG/ LawClass/LawAll.aspx?pcode=L0020176) (accessed on 28 September 2023).

Prior to conducting research, the principal investigator shall submit the research protocol for review and approval by the Institutional Review Board (hereinafter, IRB). However, the research protocol within the scope of exemption categories for IRB review, as announced by the competent authority, shall not apply.

2. Announcement of Ministry of Health and Welfare, Executive Yuan, Republic of China (Taiwan). Issue Date: 5 July 2012, Issue No.: 1010265075. (http://at.cdc.tw/N2G0v1) (accessed on 28 September 2023).

Scope of Human Research Cases Exempted from Ethics Approval Committee Review: The research case does not take minors, inmates, indigenous, pregnant women, people with physical and mental disabilities, mental patients, and other people who have been determined or judged by the review meeting to be under undue coercion or unable to make a decision of their own free will as research objects; and in any of the following circumstances below, they may be exempted from being sent to the ethics approval committee for review or be issued with a certificate of exemption by the ethics approval committee.

- 2.1 Non-anonymous, non-interactive, and non-intrusive research conducted in a public setting and in which no specific individual can be identified from the information collected.
- 2.2 Use of information that has been legally made public and use of the information conforms to the purpose of its announcement.
- 2.3 A study on the effectiveness evaluation of public policies conducted by public agencies themselves or entrusted by professional institutions when performing their statutory duties.
- 2.4 A study of educational evaluation or testing, teaching skills, or effectiveness evaluation in a general teaching environment.
- 2.5 The research project exhibits minimum risk, and the potential risk suffered by the research object is not higher than that of the researcher who did not participate in it. After evaluation by the ethics approval committee, it can be exempted from review and be issued a certificate of exemption. The minimum risk mentioned above refers to the probability or intensity of harm or discomfort suffered by the research object that is not higher than that suffered in daily life.

#### References

- 1. Kalache, A.; Keller, I. The greying world: A challenge for the twenty-first century. Sci. Prog. 2000, 83, 33–54.
- Lutz, W.; Butz, W.P.; Samir, K.C. World Population and Human Capital in the Twenty-First Century: An Overview; Oxford University Press: Oxford, UK, 2017.
- Oró-Piqueras, M.; Falcus, S. Approaches to old age: Perspectives from the twenty-first century. *Eur. J. Engl. Stud.* 2018, 22, 1–12. [CrossRef]
- Taiwan News. Taiwan Has Lowest Birth Rate in World: CIA. Available online: https://www.taiwannews.com.tw/en/news/41 80941 (accessed on 2 October 2022).
- 5. Flood, V.M. Food, nutrition and ageing in the twenty-first century. Nutr. Diet. 2018, 75, 3–5. [CrossRef]
- 6. Zheng, Z.; Chen, H. Age sequences of the elderly' social network and its efficacies on well-being: An urban-rural comparison in China. *BMC Geriatr.* **2020**, *20*, 372. [CrossRef]
- Zimmer, Z.; McDaniel, S.A. Global Ageing in the Twenty-First Century: Challenges, Opportunities and Implications; Routledge: London, UK, 2016. [CrossRef]
- 8. National Institute on Aging. Social Isolation, Loneliness in Older People Pose Health Risks. Available online: https://www.nia. nih.gov/news/social-isolation-loneliness-older-people-pose-health-risks (accessed on 2 October 2022).
- 9. Benson, J.J.; Sloan, S.; Halt, A.K. Human relationships and healthy aging. *Healthy Aging Complet. Guide Clin. Manag.* 2019, 319–328.
- 10. Fried, L.P. Designing a new social infrastructure to combat loneliness in aging adults. *Generations* 2020, 44, 1–12.
- 11. Sparling, P.B.; Howard, B.J.; Dunstan, D.W.; Owen, N. Recommendations for physical activity in older adults. *BMJ Health Care Inform.* **2015**, *350*, h100. [CrossRef] [PubMed]
- 12. Wheatley, D.; Bickerton, C. Subjective well-being and engagement in arts, culture and sport. J. Cult. Econ. 2017, 41, 23–45. [CrossRef]

- Sala, G.; Jopp, D.; Gobet, F.; Ogawa, M.; Ish-ioka, Y.; Masui, Y.; Inagaki, H.; Nakagawa, T.; Yasumoto, S.; Ishizaki, T.; et al. The impact of leisure activities on older adults' cognitive function, physical function, and mental health. *PLoS ONE*. 2019, 14, e0225006. [CrossRef]
- 14. Yoon, H.; Huber, L.; Kim, C. Sustainable aging and leisure behaviors: Do leisure activities matter in aging well? *Sustainability* **2021**, *13*, 2348. [CrossRef]
- 15. Kang, K.K.; Kim, J.A.; Kim, D. Development of a sensory gate–ball game system for the aged people. *Vis. Comput.* **2009**, *25*, 1073–1083. [CrossRef]
- Al-Adwan, A.S.; Li, N.; Al-Adwan, A.; Abbasi, G.A.; Albelbisi, N.A.; Habibi, A. Extending the technology acceptance model (TAM) to predict university students' intentions to use metaverse-based learning platforms. *Educ. Inf. Technol.* 2023, 1–33. [CrossRef]
- 17. Alfadda, H.A.; Mahdi, H.S. Measuring students' use of zoom application in language course based on the technology acceptance model (TAM). *J. Psycholinguist. Res.* **2021**, *50*, 883–900. [CrossRef] [PubMed]
- Kamal, S.A.; Shafiq, M.; Kakria, P. Investigating acceptance of telemedicine services through an extended technology acceptance model (TAM). *Technol. Soc.* 2020, 60, 101212. [CrossRef]
- Oyman, M.; Bal, D.; Ozer, S. Extending the technology acceptance model to explain how perceived augmented reality affects consumers' perceptions. *Comput. Hum. Behav.* 2022, 128, 107127. [CrossRef]
- Davis, F.D.; Bagozzi, R.P.; Warshaw, P.R. User acceptance of computer technology: A comparison of two theoretical models. *Manag. Sci.* 1989, 35, 982–1003. [CrossRef]
- 21. Cui, L.; Wang, Y.; Chen, W.; Wen, W.; Han, M.S. Predicting determinants of consumers' purchase motivation for electric vehicles: An application of Maslow's hierarchy of needs model. *Energy Policy* **2021**, *151*, 112167. [CrossRef]
- Yang, Y.; Yu, X.; Zhang, Z.; Gan, L. Integrating technology acceptance model with Maslow's hierarchy needs theory to investigate smart homes adoption. *IEEE Access* 2023, 11, 80726–80740. [CrossRef]
- Dohlman, L.; DiMeglio, M.; Hajj, J.; Laudanski, K. Global brain drain: How can the Maslow theory of motivation improve our understanding of physician migration? *Int. J. Environ. Res. Public Health* 2019, 16, 1182. [CrossRef]
- 24. Houghton, D.; Pressey, A.; Istanbulluoglu, D. Who needs social networking? An empirical enquiry into the capability of Facebook to meet human needs and satisfaction with life. *Comput. Hum. Behav.* **2020**, *104*, 106153. [CrossRef]
- 25. Maslow, A.H. A theory of human motivation. Psychol. Rev. 1943, 50, 370–396. [CrossRef]
- 26. Chiponde, D.B.; Gledson, B.; Greenwood, D. Organisational learning from failure and the needs-based hierarchy of project-based organisations. *Front. Eng. Built Environ.* **2022**, *2*, 121–132. [CrossRef]
- 27. Taiwan Ministry of Health and Welfare. 2021 Investigation Report on the Condition of the Seniors. Available online: https://dep.mohw.gov.tw/DOS/lp-5095-113.html (accessed on 2 October 2022).
- Yang, S.; Ye, S.; Li, H. Comparison of senior leisure activities in China and the United States from the perspective of cultural differences. Wirel. Commun. Mob. Comput. 2022, 2022, 8430490. [CrossRef]
- 29. Li, J.; Zeng, B.; Li, P. The influence of leisure activity types and involvement levels on leisure benefits in older adults. *Front. Public Health* **2021**, *9*, 659263. [CrossRef]
- Andrieieva, O.; Hakman, A.; Kashuba, V.; Vasylenko, M.; Patsaliuk, K.; Koshura, A.; Istyniuk, I. Effects of physical activity on aging processes in elderly persons. J. Phys. Educ. Sport 2019, 19, 1308–1314.
- 31. Duncan, A.R.C. The Concept of Leisure; Industrial Relations Centre, Queen's University: Kingston, Canada, 1963.
- Henderson, K.A. A paradox of sport management and physical activity interventions. Sport Manag. Rev. 2009, 12, 57–65. [CrossRef]
- 33. Singh, B.; Kiran, U.V. Recreational activities for senior citizens. IOSR J. Humanit. Soc. Sci. 2014, 19, 24–30. [CrossRef]
- 34. Kourkouta, L.; Koukourikos, K.; Iliadis, C.; Ouzounakis, P.; Tsaloglidou, A. Leisure time and health. *Int. J. Health Adm. Educ. Congr.* **2018**, *4*, 17–22.
- Kim, J.; Yamada, N.; Heo, J.; Han, A. Health benefits of serious involvement in leisure activities among older Korean adults. Int. J. Qual. Stud. Health Well-Being 2014, 9, 24616. [CrossRef]
- 36. Evans, W.J. Exercise training guidelines for the elderly. Med. Sci. Sports Exerc. 1999, 31, 12–17. [CrossRef]
- Venkatesh, V.; Davis, F.D. A theoretical extension of the technology acceptance model: Four longitudinal field studies. *Manag. Sci.* 2000, 46, 186–204. [CrossRef]
- Gayman, A.M.; Fraser-Thomas, J.; Dionigi, R.A.; Horton, S.; Baker, J. Is sport good for older adults? A systematic review of psychosocial outcomes of older adults' sport participation. *Int. Rev. Sport Exerc. Psychol.* 2017, 10, 164–185. [CrossRef]
- The Croquet Association. The Basic Rules of Golf Croquet. Available online: https://www.croquet.org.uk/?p=games/golf/ rules/gcBasicRules (accessed on 2 October 2022).
- 40. Britannica, T. (Ed.) *Argon Encyclopedia Britannica*; Association Croquet: Cheltenham, UK, 2022; Available online: https://www. britannica.com/sports/association-croquet (accessed on 2 October 2022).
- 41. Wikipedia. Croquet. Available online: https://en.wikipedia.org/wiki/Croquet (accessed on 2 October 2022).
- 42. Shah, J.L.; Bhat, H.F.; Khan, A.I. Integration of cloud and IoT for smart e-healthcare. *Healthc. Paradig. Internet Things Ecosyst.* 2021, *6*, 101–136.
- 43. Dhatterwal, J.S.; Kaswan, K.S.; Baliyan, A.; Jain, V. Integration of cloud and IoT for smart e-healthcare. In *Connected e-Health*, *Studies in Computational Intelligence*; Springer: Cham, Switzerland, 2022; Volume 1021, pp. 1–31.

- 44. Shukla, S.; Hassan, M.F.; Khan, M.K.; Jung, L.T.; Awang, A. An analytical model to minimize the latency in healthcare Internet-of-Things in fog computing environment. *PLoS ONE* **2019**, *14*, e0224934. [CrossRef]
- 45. Kumar, S.; Tiwari, P.; Zymbler, M. Internet of Things is a revolutionary approach for future technology enhancement: A review. *J. Big Data* **2019**, *6*, 111. [CrossRef]
- Nižetić, S.; Šolić, P.; González-De, D.L.D.I.; Patrono, L. Internet of Things (IoT): Opportunities, issues and challenges towards a smart and sustainable future. J. Clean Prod. 2020, 274, 122877. [CrossRef]
- 47. Borelli, E.; Paolini, G.; Antoniazzi, F.; Barbi-roli, M.; Benassi, F.; Chesani, F.; Chiari, L.; Fantini, M.; Fuschini, F.; Galassi, A.; et al. HABITAT: An IoT solution for independent elderly. *Sensors* **2019**, *19*, 1258. [CrossRef]
- Nord, J.H.; Koohang, A.; Paliszkiewicz, J. The Internet of Things: Review and theoretical framework. *Expert Syst. Appl.* 2019, 133, 97–108. [CrossRef]
- Čolaković, A.; Hadžialić, M. Internet of Things (IoT): A review of enabling technologies, challenges, and open research issues. Comput. Netw. 2018, 144, 17–39. [CrossRef]
- Qian, K.; Zhang, Z.; Yamamoto, Y.; Schuller, B.W. Artificial intelligence internet of things for the elderly: From assisted living to healthcare monitoring. *IEEE Signal Process. Mag.* 2021, *38*, 78–88. [CrossRef]
- Abdi, S.; Witte, L.D.; Hawley, M. Exploring the potential of emerging technologies to meet the care and support needs of older people: A delphi survey. *Geriatrics* 2021, 6, 19. [CrossRef] [PubMed]
- Nagajayanthi, B. Decades of Internet of Things towards twenty-first century: A re-search-based introspective. Wirel. Pers. Commun. 2022, 123, 3661–3697. [CrossRef] [PubMed]
- El Khaddar, M.A.; Boulmalf, M. Smartphone: The Ultimate IoT and IoE Device. In Smartphones from an Applied Research Perspective, 2nd ed.; Mohamudally, N., Ed.; Books on Demand GmbH: Norderstedt, Germany, 2017; pp. 137–200.
- 54. Coghlan, S.; Waycott, J.; Lazar, A.; Bar-bosa Neves, B. Dignity, autonomy, and style of company: Dimensions older adults consider for robot companions. *Proc. ACM Hum.-Comput. Interact.* **2021**, *5*, 1–25. [CrossRef] [PubMed]
- Bradwell, H.L.; Edwards, K.J.; Winnington, R.; Thill, S.; Jones, R.B. Companion robots for older people: Importance of user-centred design demonstrated through observations and focus groups comparing preferences of older people and roboticists in South West England. *BMJ Open* 2019, *9*, e032468. [CrossRef]
- 56. Payr, S.; Werner, F.; Werner, K. Potential of Robotics for Ambient Assisted Living; FFG Benefit: Vienna, Austria, 2015.
- 57. Pu, L.; Moyle, W.; Jones, C.; Todorovic, M. The effectiveness of social robots for older adults: A systematic review and metaanalysis of randomized controlled studies. *Gerontologist* **2019**, *59*, e37–e51. [CrossRef] [PubMed]
- Robinson, N.L.; Cottier, T.V.; Kavanagh, D.J. Psychosocial Health Interventions by Social Robots: Systematic Review of Randomized Controlled Trials. J. Med. Internet. Res. 2019, 21, e13203. [CrossRef]
- 59. Hung, L.; Liu, C.; Woldum, E.; Au-Yeung, A.; Berndt, A.; Wallsworth, C.; Chaudhury, H. The benefits of and barriers to using a social robot PARO in care settings: A scoping review. *BMC Geriatr.* **2019**, *19*, 232. [CrossRef]
- 60. Wachsmuth, I. Robots like me: Challenges and ethical issues in aged care. Front. Psychol. 2018, 9, 432. [CrossRef]
- 61. Rong, G.; Mendez, A.; Assi, E.B.; Zhao, B.; Sawan, M. Artificial intelligence in healthcare: Review and prediction case studies. *Engineering* **2020**, *6*, 291–301. [CrossRef]
- 62. Shah, M.U.; Iqbal, F. Assisting People of Determination and the Elderly Using Social Robot: A Case Study. In Proceedings of the 55th Hawaii International Conference on System Sciences, Maui, HI, USA, 4–7 January 2022.
- 63. Ofcom. 2022 Adults' Media Use and Attitudes Report. Available online: https://www.ofcom.org.uk/\_\_data/assets/pdf\_file/00 20/234362/adults-media-use-and-attitudes-report-2022.pdf (accessed on 2 October 2022).
- 64. Kim, S.; Yao, W.; Du, X. Exploring older adults' adoption and use of a tablet computer during COVID-19: Longitudinal qualitative study. *JMIR Aging* **2022**, *5*, e32957. [CrossRef]
- 65. AT Kearney. Understanding the Needs and Consequences of the Aging Consumer. Available online: https://www.theconsumergoodsforum.com/wp-content/uploads/2017/11/CGF-Understanding-the-Needs-of-ageing\_consumer\_report.pdf (accessed on 2 October 2022).
- InsightXplorer. 2013 Behavior Report of Using Smart Phones. Available online: https://www.bnext.com.tw/article/27131/BN-ARTICLE-27131 (accessed on 2 October 2022).
- 67. Dennison, L.; Morrison, L.; Conway, G.; Yardley, L. Opportunities and challenges for smartphone applications in supporting health behavior change: Qualitative study. *J. Med. Internet Res.* **2013**, *15*, e2583. [CrossRef]
- Eshet, E.; Bouwman, H. Addressing the context of use in mobile computing: A survey on the state of the practice. *Interact. Comput.* 2015, 27, 392–412. [CrossRef]
- 69. Barkley, J.E.; Lepp, A.; Santo, A.; Glickman, E.; Dowdell, B. The relationship between fitness app use and physical activity behavior is mediated by exercise identity. *Comput. Hum. Behav.* **2020**, *108*, 106313. [CrossRef]
- Herrmann, L.K.; Kim, J. The fitness of apps: A theory-based examination of mobile fitness app usage over 5 months. *Mhealth* 2017, 3, 28293619. [CrossRef] [PubMed]
- Muntaner-Mas, A.; Martinez-Nicolas, A.; Lavie, C.J.; Blair, S.N.; Ross, R.; Arena, R.; Ortega, F.B. A systematic review of fitness apps and their potential clinical and sports utility for objective and remote assessment of cardiorespiratory fit-ness. *Sports Med.* 2019, 49, 587–600. [CrossRef]
- 72. Cai, J.; Zhao, Y.; Sun, J. Factors influencing fitness app users' behavior in China. *Int. J. Hum.-Comput. Interact.* **2022**, *38*, 53–63. [CrossRef]

- Liao, C.C.; Hwang, D.H.; Koike, H. AI Golf: Golf Swing Analysis Tool for Self-Training. *IEEE Access* 2022, 10, 106286–106295. [CrossRef]
- Wang, S.; Xu, Y.; Zheng, Y.; Zhu, M.; Yao, H.; Xiao, Z. Tracking a golf ball with high-speed stereo vision system. *IEEE Trans. Instrum. Meas.* 2018, 68, 2742–2754. [CrossRef]
- 75. Strava. Clubs on Strava. Available online: https://www.strava.com/ (accessed on 2 October 2022).
- 76. Franken, R.; Bekhuis, H.; Tolsma, J. Kudos make you run! How runners influence each other on the online social network Strava. *Soc. Netw.* **2023**, *72*, 151–164. [CrossRef]
- Auto Scoring System for Golf Strike Back Ball. R. O. C. Patent TWM602930, In Intellectual Property Office, Ministry of Economic Affairs, R.O.C. R. Available online: https://twpat.tipo.gov.tw/ (accessed on 2 October 2022).
- 78. Yousafzai, S.Y.; Foxall, G.R.; Pallister, J.G. Technology acceptance: A meta-analysis of the TAM: Part 1. *J. Model. Manag.* 2007, 2, 251–280. [CrossRef]
- 79. Yousafzai, S.Y.; Foxall, G.R.; Pallister, J.G. Technology acceptance: A meta-analysis of the TAM: Part 2. *J. Model. Manag.* 2007, 2, 281–304. [CrossRef]
- Legris, P.; Ingham, J.; Collerette, P. Why do people use information technology? A critical review of the technology acceptance model. *Inf. Manag.* 2003, 40, 191–204. [CrossRef]
- Abdul Hamid, H.H.; Ahmad Sherjawi, Z.; Omar, S.; Phon-Amnuaisuk, S. Student acceptance and attitude towards using 3D virtual learning spaces. In *Computational Intelligence in Information Systems: Proceedings of the Computational Intelligence in Information Systems Conference (CIIS 2016), Brunei, Brunei Darussalam, 18–20 November 2016; Phon-Amnuaisuk, S., Au, T.W., Omar,* S., Eds.; Springer: Berlin/Heidelberg, Germany, 2017; Volume 532, pp. 107–118.
- 82. Shuhaiber, A.; Mashal, I. Understanding users' acceptance of smart homes. Technol. Soc. 2019, 58, 101110. [CrossRef]
- 83. Schill, M.; Godefroit-Winkel, D.; Diallo, M.F.; Barbarossa, C. Consumers' intentions to purchase smart home objects: Do environmental issues matter? *Ecol. Econ.* 2019, *161*, 176–185. [CrossRef]
- Tichenor, P.J.; Donohue, G.A.; Olien, C.N. Mass media flow an differential growth in knowledge. *Public Opin. Q.* 1970, 34, 159–170. [CrossRef]
- Youn, S.Y.; Lee, K.H. Proposing value-based technology acceptance model: Testing on paid mobile media service. *Fash. Text.* 2019, 6, 13. [CrossRef]
- Peng, M.Y.P.; Yan, X. Exploring the influence of determinants on behavior intention to use of multiple media kiosks through technology readiness and acceptance model. *Front. Psychol.* 2022, 13, 852394. [CrossRef]
- 87. Hsu, H.H.; Chang, Y.Y. Extended TAM model: Impacts of convenience on acceptance and use of Moodle. *Online Submiss.* **2013**, *3*, 211–218.
- 88. Wang, C.; Ahmad, S.F.; Ayassrah, A.Y.B.A.; Awwad, E.M.; Irshad, M.; Ali, Y.A.; AI-Razgan, M.; Khan, Y.; Han, H. An empirical evaluation of technology acceptance model for Artificial Intelligence in E-commerce. *Heliyon* **2023**, *9*, e18349. [CrossRef]
- Carducci, B.J. Maslow's Hierarchy of Needs. In *The Wiley Encyclopedia of Personality and Individual Differences: Models and Theories*; Wiley: New York, NY, USA, 2020; pp. 269–273.
- 90. Yang, K. Managing Human Behavior in Public and Nonprofit Organizations. Public Adm. Q. 2001, 25, 518–521.
- Ryan, B.J.; Coppola, D.; Canyon, D.V.; Brickhouse, M.; Swienton, R. COVID-19 community stabilization and sustainability framework: An integration of the Maslow hierarchy of needs and social determinants of health. *Dis. Med. Public Health Prep.* 2020, 14, 623–629. [CrossRef] [PubMed]
- 92. Altymurat, A.; Muhai, M.; Saparow, T. Human Behavior in Organizations Related to Abraham Maslow's Hierarchy of Needs Theory. *Interdiscip. J. Pap. Hum. Rev.* 2021, 2, 12–16. [CrossRef]
- Hale, A.J.; Ricotta, D.N.; Freed, J.; Smith, C.C.; Huang, G.C. Adapting Maslow's hierarchy of needs as a framework for resident wellness. *Teach. Learn. Med.* 2019, 31, 109–118. [CrossRef] [PubMed]
- Orth, U.; Trzesniewski, K.H.; Robins, R.W. Self-esteem development from young adulthood to old age: A cohort-sequential longitudinal study. J. Pers. Soc. Psychol. 2010, 98, 645–658. [CrossRef]
- 95. Bhattacherjee, A. Understanding information systems continuance: An expectation confirmation model. *MIS Q.* **2001**, *25*, 351–370. [CrossRef]
- 96. Sørebø, Ø.; Eikebrokk, T.R. Explaining IS continuance in environments where usage is mandatory. *Comput. Hum. Behav.* 2008, 24, 2357–2371. [CrossRef]
- 97. Van Eckert, S.; Gaidys, U.; Martin, C.R. Self-esteem among German nurses: Does academic education make a difference? *J. Psychiatr. Ment. Health Nurs.* **2012**, *19*, 903–910. [CrossRef]
- 98. Tolstedt, B.E.; Stokes, J.P. Relation of verbal, affective, and physical intimacy to marital satisfaction. *J. Couns. Psychol.* **1983**, *30*, 573–580. [CrossRef]
- 99. Sinclair, V.G.; Dowdy, S.W. Development and validation of the emotional intimacy scale. *J. Nurs. Meas.* 2005, 13, 193–206. [CrossRef]
- 100. Ridings, C.M.; Gefen, D. Virtual community attraction: Why people hang out online. *J. Comput.-Mediat. Commun.* **2004**, *10*, JCMC10110. [CrossRef]
- 101. Rau, P.L.P.; Gao, Q.; Ding, Y.N. Relationship between the level of intimacy and lurking in online social network services. *Comput. Hum. Behav.* **2008**, *24*, 2757–2770. [CrossRef]

- 102. Lu, J.; Yao, J.E.; Yu, C.S. Personal innovativeness, social influences and adoption of wireless internet services via mobile technology. *J. Strateg. Inf. Syst.* **2005**, *14*, 245–268. [CrossRef]
- 103. Bi, N.C.; Zhang, R. "I will buy what my 'friend' recommends": The effects of parasocial relationships, influencer credibility and self-esteem on purchase intentions. *J. Res. Interact. Mark.* **2022**, *17*, 157–175. [CrossRef]
- 104. Turaga, R.M.; Howarth, R.B.; Borsuk, M.E. Pro-environmental behavior: Rational choice meets moral motivation. *Ann. N. Y. Acad. Sci.* **2010**, *1185*, 24–211. [CrossRef]
- Lin, C.L.; Lee, S.H.; Horng, D.J. The effects of online reviews on purchasing intention: The moderating role of need for cognition. Soc. Behav. Personal. Int. J. 2011, 39, 71–81. [CrossRef]
- 106. Hussain, A.; Shabir, G. Cognitive needs and use of social media: A comparative study of gratifications sought and gratification obtained. *Inf. Discov. Deliv.* 2020, *48*, 79–90. [CrossRef]
- 107. Zhao, J.; Zhu, C. Modeling and quantifying the impact of personified communication on purchase behavior in social commerce. *Behav. Sci.* **2023**, *13*, 627. [CrossRef]
- 108. Cyr, D.; Head, M.; Ivanov, A. Design aesthetics leading to m-loyalty in mobile commerce. Inf. Manag. 2006, 43, 950–963. [CrossRef]
- 109. Liao, C.; To, P.; Liu, C.; Kuo, P.; Chuang, S. Factors influencing the intended use of web portals. Online Inf. Rev. 2011, 35, 237–254. [CrossRef]
- 110. Jeong, J.; Kim, Y.; Roh, T. Do consumers care about aesthetics and compatibility? The intention to use wearable de-vices in health care. *SAGE Open* **2022**, *11*, 215824402110400. [CrossRef]
- Tseng, P.Y.; Lee, S.F. The impact of web visual aesthetics on purchase intention. In Proceedings of the 2019 IEEE Eurasia Conference on IOT, Communication and Engineering (ECICE), Yunlin, Taiwan, 3–6 October 2019; pp. 28–31.
- 112. Denison-Day, J.L.; Muir, S.; Newell, C.; Appleton, K.M. The Role of Aesthetics in Intentions to Use Digital Health Interventions. *PLOS Digit. Health* **2023**, *2*, e0000274. [CrossRef] [PubMed]
- 113. Derlega, V.J.; Chaikin, A.L. Privacy and self-disclosure in social relationships. J. Soc. Issue 1977, 33, 102–115. [CrossRef]
- 114. Daimi, S.; Tolunay, A. An empirical investigation on influencer marketing: The impact of content-related, follower-related and influencer-related factors on consumers' purchase intentions. *Istanb. Manag. J.* **2021**, *91*, 59–86. [CrossRef]
- Huta, V.; Ryan, R.M. Pursuing pleasure or virtue: The differential and overlapping well-being benefits of hedonic and eudaimonic motives. J. Happiness Stud. 2010, 11, 735–762. [CrossRef]
- Huang, C.H. Study on the Application of Technology Acceptance Model to Community Remote Care System. Master's Thesis, Nan Kai University of Technology, Caotun, Taiwan, 8 July 2010.
- 117. Kuo, Y.Y. A Research on the Influence on Customers' Use Intention of Electronic Vehicles with the Technology Acceptance Model. Master's Thesis, National Cheng Kung University, Tainan City, Taiwan, 13 June 2013.
- 118. Tang, K.M. The Study of Probing into Aborigine Adult e-Learning with Technology Acceptance Model-Take "Aborigine Cyber Institute" for Instance. Master's Thesis, National Chi Nan University, Puli, Taiwan, March 2008.
- 119. Hsu, P.Y. The Research on the Middle-Aged and Older People Regarding the Importance and Satisfaction with Maslow's Hierarchy of Needs. Master's Thesis, Nan Kai University of Technology, Caotun, Taiwan, 12 June 2015.
- Chen, Y.L. A Study of the Religious Activities in Rural Taiwan Basing on Maslow's Hierarchy Needs and the Leisure Theory. Master's Thesis, Ling Tung University of Technology, Taichung City, Taiwan, 15 July 2020.
- 121. Mueller, R.O.; Hancock, G.R. Factor Analysis and Latent Structure, Confirmatory. In *International Encyclopedia of the Social & Behavioral Sciences*; Elsevier: Amsterdam, the Netherlands, 2001; pp. 5239–5244.
- 122. Aguinis, H. Statistical power problems with moderated multiple regression in management research. *J. Manag.* **1995**, *21*, 1141–1158. [CrossRef]
- 123. Aguinis, H.; Pierce, C.A.; Bosco, F.A.; Muslin, I.S. First decade of Organizational Research Methods: Trends in design, measurement, and data-analysis topics. *Organ. Res. Methods* **2009**, *12*, 69–112. [CrossRef]
- 124. Fisher, R.A. Studies in crop variation. I. An examination of the yield of dressed grain from Broadbalk. J. Agric. Sci. 1921, 11, 107–135. [CrossRef]
- 125. Banerjee, A.; Chitnis, U.B.; Jadhav, S.L.; Bhawalkar, J.S.; Chaudhury, S. Hypothesis testing, type I and type II errors. *Ind. Psychiatry* J. 2009, 18, 127–131. [CrossRef]
- 126. Comrey, A.L. A First Course in Factor Analysis; Academic Press: New York, NY, USA, 1973.
- 127. Gorsuch, R.L. Factor Analysis; Lawrence Erlbaum: Hillsdale, NJ, USA, 1983.
- 128. Tinsley, H.E.; Tinsley, D.J. Uses of factor analysis in counseling psychology. J. Couns. Psychol. 1987, 34, 414–424. [CrossRef]
- 129. Comrey, A.L. Factor analytic methods of scale development in personality and clinical psychology. *J. Consult. Clin. Psychol.* **1988**, 56, 754–761. [CrossRef]
- 130. Kaiser, H.F. The varimax criterion for analytic rotation in factor analysis. *Psychometrika* 1958, 23, 187–200. [CrossRef]
- 131. Guilford, J.P. Fundamental Statistics in Psychology and Education, 2nd ed.; McGraw-Hill: New York, NY, USA, 1950.
- 132. Cronbach, L.J. Coefficient alpha and the internal structure of tests. Psychometrika 1951, 16, 297–334. [CrossRef]
- 133. Kerlinger, F.N. Foundations of Behavioral Research; Holt, Rinehart and Winston: New York, NY, USA, 1966.
- 134. Kaiser, H.F. A second generation little jiffy. *Psychometrika* **1970**, *35*, 401–415. [CrossRef]
- 135. Kaiser, H.F.; Rice, J. Little jiffy, mark IV. Educ. Psychol. Meas. 1974, 34, 111-117. [CrossRef]
- 136. Srinivasan, A. Alternative measures of system effectiveness: Associations and implications. MIS Q. 1985, 9, 243–253. [CrossRef]

- 137. Ginzberg, M.J. Early diagnosis of MIS implementation failure: Promising results and unanswered questions. *Manag. Sci.* **1981**, 27, 459–478. [CrossRef]
- 138. Davis, F.D.; Venkatesh, V. Toward preprototype user acceptance testing of new information systems: Implications for soft-ware project management. *IEEE Trans. Eng. Manag.* 2004, *51*, 31–46. [CrossRef]
- 139. Jarupunphol, P.; Buathong, W. Potential Use of Smart Kitchen Management Application (SKM). In Proceedings of the 2018 22nd International Computer Science and Engineering Conference (ICSEC), Chiang Mai, Thailand, 21–24 November 2018.
- 140. Anderson, J.R. The Architecture of Cognition; Psychology Press: Hove, UK, 1996.
- 141. Fang, Y.H. Does online interactivity matter? Exploring the role of interactivity strategies in consumer decision making. *Comput. Hum. Behav.* **2012**, *28*, 1790–1804. [CrossRef]
- 142. Yim, M.Y.C.; Chu, S.C.; Sauer, P.L. Is augmented reality technology an effective tool for e-commerce? An interactivity and vividness perspective. J. Interact. Mark. 2017, 39, 89–103. [CrossRef]
- Liao, G.Y.; Pham, T.T.L.; Cheng, T.C.E.; Teng, C.I. Impacts of real-world need satisfaction on online gamer loyalty: Perspective of self-affirmation theory. *Comput. Hum. Behav.* 2020, 103, 91–100. [CrossRef]
- 144. Csikszentmihalyi, M. Beyond Boredom and Anxiety; Jossey-bass: Hoboken, NJ, USA, 2000.
- 145. Yoon, N.; Kim, H.; Choo, H.J. The effect of shopping value on continuous use intention of online cross-border shopping mediated by curiosity and self-efficacy-comparing heavy and light user. *J. Korean Soc. Cloth. Text.* **2020**, *44*, 1004–1018. [CrossRef]
- 146. Tsai, T.H.; Chang, H.T.; Chen, Y.J.; Chang, Y.S. Determinants of user acceptance of a specific social platform for older adults: An empirical examination of user interface characteristics and behavioral intention. *PLoS ONE* 2017, 12, e0180102. [CrossRef] [PubMed]
- 147. Zhang, Y.; Dang, Y. Investigating essential factors on students' perceived accomplishment and enjoyment and intention to learn in web development. *ACM Trans. Comput. Educ.* **2015**, *15*, 1–21. [CrossRef]
- 148. Solomon, B.C.; Nikolaev, B.N.; Shepherd, D.A. Does educational attainment promote job satisfaction? The bittersweet trade-offs between job resources, demands, and stress. J. Appl. Psychol. 2022, 107, 1227. [CrossRef] [PubMed]
- 149. Yu, C.W.; Chang, C.H. The study of exploring the needs of the elderly based on institutionalization and communization. In Proceedings of the 2013 International Conference on Gerontechnology and Service Management, Nantou, Taiwan, 16 May 2013.
- 150. Malm, C.; Jakobsson, J.; Isaksson, A. Physical activity and sports—Real health benefits: A review with insight into the public health of Sweden. *Sports* **2019**, *7*, 127. [CrossRef]
- 151. Papi, S.; Cheraghi, M. Multiple factors associated with life satisfaction in older adults. *Prz. Menopauzalny* **2021**, 20, 69–75. [CrossRef]
- 152. Baird, B.M.; Lucas, R.E.; Donnellan, M.B. Life satisfaction across the lifespan: Findings from two nationally representative panel studies. *Soc. Indic. Res.* 2010, *99*, 183–203. [CrossRef]
- 153. Thorpe, H. Action sports, social media, and new technologies: Towards a research agenda. *Commun. Sport* **2017**, *5*, 554–578. [CrossRef]
- 154. Cho, H.; Chi, C.; Chiu, W. Understanding sustained usage of health and fitness apps: Incorporating the technology acceptance model with the investment model. *Technol. Soc.* **2020**, *63*, 101429. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.